	GENER	AL PUBLIC TARIFF OF APM TER	MINALS CALLA	0 S.A.							
								1			v 12.10
ltem	Services description	Nature	Unit of charge		Tarif	f (USD)			Special	tariff (USD)	
SECTION 1		CONTAINERS	i -								
Section 1.1	Standard Services	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
1.1.1	Standard Service to the Vessel - Services to the Vessel										
1.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.69	0% Rate			1.42	0% Rate		
1.1.2	Loading and Unloading Containers - STS Gantry Cranes (n2 and n3)										
1.1.2.1	Loading and Unloading Full 20 Foot Container with dry cargo		Per move	11.45	0% Rate	194.10	34.94	10.80	0% Rate	173.60	31.1
1.1.2.2	Loading and Unloading Full 40 Foot Container with dry cargo		Per move	11.45	0% Rate	331.50	59.67	10.80	0% Rate	291.70	52.
1.1.2.3	Loading and Unloading Full 20 Foot Container with refrigerated cargo		Per move	11.45	0% Rate	194.10	34.94	10.80			
1.1.2.4	Loading and Unloading Full 40 Foot Container with refrigerated cargo	Regulated	Per move	11.45	0% Rate	331.50	59.67	10.80			
1.1.2.5	Loading and Unloading Empty 20 Foot Container	regulated	Per move	168.30	0% Rate			139.00	0% Rate		
1.1.2.6	Loading and Unloading Empty 40 Foot Container		Per move	262.30	0% Rate			205.30	0% Rate		
1.1.2.7	Complete cycle 20 Foot Transhipment Container (full or empty) within the Terminal (n4)		Per complete cycle	200.00	0% Rate			119.00	0% Rate		
1.1.2.8	Complete cycle 40 Foot Transhipment Container (full or empty) within the Terminal (n4)		Per complete cycle	305.00	0% Rate			178.00	0% Rate		
1.1.3	Loading and Unloading Containers - Without STS Gantry Cranes (n3 and n5)										
1.1.3.1	Loading and Unloading Full 20 Foot Container with dry cargo		Per move	6.20	0% Rate	176.40	31.75	5.90	0% Rate	139.90	25.
1.1.3.2	Loading and Unloading Full 40 Foot Container with dry cargo		Per move	6.20	0% Rate	289.70	52.15	5.90	0% Rate	230.50	41.
1.1.3.3	Loading and Unloading Full 20 Foot Container with refrigerated cargo		Per move	6.20	0% Rate	176.40	31.75	5.90	0% Rate		
1.1.3.4	Loading and Unloading Full 40 Foot Container with refrigerated cargo	Regulated	Per move	6.20	0% Rate	289.70	52.15		0% Rate		
1.1.3.5	Loading and Unloading Empty 20 Foot Container	regulated	Per move	139.30	0% Rate			118.40	0% Rate		
1.1.3.6	Loading and Unloading Empty 40 Foot Container		Per move	217.30	0% Rate			176.20	0% Rate		
1.1.3.7	Complete cycle 20 Foot Transhipment Container (full or empty) within the Terminal (n4)		Per complete cycle	158.80	0% Rate			109.00	0% Rate		
1.1.3.8	Complete cycle 40 Foot Transhipment Container (full or empty) within the Terminal (n4)		Per complete cycle	237.30	0% Rate			163.00	0% Rate		

(iii): The containers whose dimensions are less than 20-too containers' will be charged with 20-too container tarff. The containers whose dimensions are higher than 20-too containers' will be charged with 40-too container tarff.

(1) Transhipment containers are hose which arrive and are aligned in vessels that operate in the North Terminal. The tarff includes loading/unloading in both ships (complete transhipment containers are asse, with the exception of containers with oversized cargo that are billed for use of operational area tom day one, as it is a special service. Inter-terminals Transhipment containers are also invoiced according to described tarifin in this section (1.2.5, 1.1.2.6, 1.1.3.5 y 1.1.3.6), in which tarff applies per move rather than per complete cycle. If also induces cabdage containers that are disbarged in one align and are toxed containers with STS ganity cranes (are inter-service). If evaluates two days of operative area use, with the exception of containers with the exception of containers with STS ganity cranes and are then loaded in vessels at better with STS ganity cranes (are inter-service). If evaluates two days of operative area use, with the exception of containers with STS ganity cranes and are then loaded in vessels at better with STS ganity cranes (are inter-service). If evaluates two days of operative area use, with the exception of containers with STS ganity cranes and are then loaded in vessels at better with STS ganity cranes (are inter-service). If evaluates two days of operative area use, with the exception of containers that are disclosed and then evaluated with evaluated on the evaluated on the evaluation of the

(n5): This is applicable only in those berths which do not have STS gantry cranes and in those cases when loading/unloading operations of containers will be carried out without STS gantry cranes.

(n5): This is appli	cable only in those berths which do not have STS gantry cranes and in those cases when loading/unloading operations of containers will be carried out without STS gantry cranes.									
Section 1.2	Special Services to the Vessel (n6)	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff) Car	go (IGV) Vessel (	Tariff) Vessel (IGV)	Cargo (Tariff) C	Cargo (IGV)
1.2.1	Hatches or Hold Lids (n7)									
1.2.1.1	Movement of ISO hatches	Regulated	Per move	0.00	0% Rate					
1.2.2	Restow of containers (n6)									
1.2.2.1	Restow same hold 20 foot standard container		Per container	0.00	0% Rate					
1.2.2.2	Restow same hold 40 foot standard container		Per container	0.00	0% Rate					
1.2.2.3	Restow via guay 20 foot standard container	Regulated	Per container	0.00	0% Rate					
1.2.2.4	Restow via quay 40 foot standard container		Per container	0.00	0% Rate					
1.2.3	Increase of container productivity with Mobile Harbor Cranes (MHC)									
1.2.3.1	Supply of MHC to increase productivity during loading/unloading container operations (price per shift or shift fraction) (n8)	Non Regulated	Per Crane/Shift	2.500.00	0% Rate					
1.2.4	Use of Operational Area - Import Empty Containers (except Transhipment)	-								
1.2.4.1	48 hours (Free time included in Standard Service) (n9)	Regulated		Fre	e					
1.2.4.2	Day: 3 - 6 (Tariff per day)		Per TEU/day	10.70	0% Rate					
1.2.4.3	Day 7 (Price per day)	Non Regulated	Per TEU/day	37.50	0% Rate					
1.2.4.4	Day: 8 onwards (Price per day)		Per TEU/day	16.10	0% Rate					
1.2.5	Use of Operational Area - Export Empty Containers (except Transhipment)		T di TEO/day	10.10	01011010					
1.2.5.1	Z hours (Free time) (n10)	Regulated		Fre	e					
1.2.5.2	Days: 4 – 6 (Tariff per day)	rogalated	Per TEU/day	10.70	0% Rate					
1.2.5.2	Days: + - 0 (fainin per day) Day 7 (Price per day)	Non Regulated	Per TEU/day	37.50	0% Rate					
1.2.5.4	Day: 8 onwards (Price per day)		Per TEU/day	16.10	0% Rate					
1.2.5.4	Use of Operational Area - Transhipment Full Containers		Fel TEU/uay	10.10	U/6 Public					
1.2.6.1	48 hours (Free time included in Standard Service) (n11)	Regulated		Fre						
1.2.6.1		regulated	Per TEU/day	17.10	0% Rate					
	Day: 3 – 4 (Price per day)	New Developed								
1.2.6.3	Day: 5 – 6 (Price per day)	Non Regulated	Per TEU/day	26.80	0% Rate					
1.2.6.4	Day 7 onwards (Price per day)		Per TEU/day	33.00	0% Rate					
1.2.7	Use of Operational Area - Transhipment Empty Containers									
1.2.7.1	48 hours (Free time included in Standard Service) (n11)	Regulated		Fre						
1.2.7.2	Day: 3 – 4 (Price per day)		Per TEU/day	17.10	0% Rate					
1.2.7.3	Day: 5 – 6 (Price per day)	Non Regulated	Per TEU/day	26.80	0% Rate					
1.2.7.4	Day 7 onwards (Price per day)		Per TEU/day	33.00	0% Rate					
1.2.8	Use of Operational Area for Empty pool - Empty containers (except Transhipment)									
1.2.8.1	Use of operational area for empty containers from empty pool (n12)	Non Regulated	Per TEU	Per agre	ement					
1.2.9	Special Container - empty and transhipment									
1.2.9.1	Loading/unloading and delivery of non ISO/OOG (Additional) Container 20 and 40 foot (n7)	Regulated	Per container	0.00	0% Rate					
1.2.9.2	Treatment of Hazardous cargo containers IMO class 1 (n13)		Per TEU	260.00	0% Rate					
1.2.9.3	Treatment of Hazardous cargo containers IMO class 5.2, class 6.2 and class 7 (n13)		Per TEU	260.00	0% Rate					
1.2.9.4	Treatment of Hazardous cargo IMO 9 (n13)		Per TEU	120.00	0% Rate					
1.2.9.5	Treatment of Hazardous cargo other IMO classes (n13)	Non Regulated	Per TEU	150.00	0% Rate					
1.2.9.6	Supply of special equipment for handling 20' containers with oversized cargo in the container yard	Non Regulated	Container-Move	169.00	0% Rate					
1.2.9.7	Supply of special equipment for handling 40' containers with oversized cargo in the container yard		Container-Move	330.00	0% Rate					
1.2.9.8	Containers with oversized cargo - use of operational area (n14)		Per container/day	25.00	0% Rate					
1.2.9.9	High Cube Containers 96" (n14)		Per container	20.60	0% Rate					
1.2.10	Complementary Services to stevedoring/unstevedoring									
1.2.10.1	Connection and disconnection service for reefer containers on-board vessels (n15)	Non Regulated	Per container-operation	9.00	0% Rate					
1.2.11	Transhioment Reefer Containers									
1.2.11.1	Energy - container 20 and 40 foot (n6 y n16)		Per Container/hour	0.00	0% Rate					
1.2.11.2	Inspection and monitoring - container 20 and 40 foot (n7 and n16)	Regulated	Per Container/day	0.00	0% Rate					
1.2.11.3	Energy - container 20 and 40 foot (n17)		Per Container/day	75.00	0% Rate					
1.2.11.4	Inspection and monitoring - container 20 and 40 foot (n17)	Non Regulated	Per event	27.20	0% Rate					
1.2.12	Re-stow Refer Containers (n18)			2.120						
1.2.12.1	Energy supply for re-stow refere containers	Non Regulated	Per container	77.10	0% Rate					
1.2.13	Casification of a full container reefer - all traffics (n19)		i or contained		010100					
1.2.13.1	Supply of gases N2 y CO2		Per container	373.89	67.30					
1.2.13.1	Supply of gases N2 y CO2 Supply of gases N2	Regulated	Per container	333.56	60.04					
1.2.13.2	Supply of gases CO2	. aguiated	Per container	133.50	24.03					
1.2.13.3	Supply of gases CO2 Reefer container data download		rei container	133.50	24.03					
1.2.14		Regulated	Per container	21.22	3.82					
	Download and sending information of reefer container parameters	regulated	rei container	21.22	3.62					
1.2.15	Additional inspection of seals	New Developed	Descent	40.00	Of Data					
1.2.15.1	Additional inspection or revision of seals (n20)	Non Regulated	Per seal	12.00	0% Rate					
1.2.16	Traction Service for Inter-terminal Transhipment Containers			10.00						
1.2.16.1	Inter-terminal Transport of Transhipment Containers (n21)	Non Regulated	Per container	40.00	0% Rate					
(no): Al the other	services applying to all types of vessels are available in section 7 of this Tariff Schedule.									

(nf): All the other services applying to all types of vessels are available in section 7 of this Tarff Schedule. (nf): Applicable to all users that necessarily require the service), according to conditions established in the Tarff. Regulation and Commercial Policy of APM Terminals Calao S.A. (nf): Charge of this price is applicable whenever service is required by users and is almed to increase productivity of loading/unicading container operations above productivity levels set furth in the Concession Agreement. This price is additional to standard service without ganty crane. In case productivity is increased by initiative of APM Terminals Calao S.A. and with no user request, productivity increase cost is under responsability of APM Terminals Calao. Arm remines used. (n10): The fine tho can as from the end of complete disabarge operation of the ship. (n10): 72 Free hours to run as from the period cargo is located in the Terminal yeard for shipment. Storage time ends with vessel ETB sent to the users. For purposes of computation of storage free time storage and billable storage, it should be taken in consideration is the so-called 'Billable ETB' that is published in the "Container Ship Programme" option on our website and defined in the Tariffs Regulation and Commercial Policy of APM (n10): 72 Free hours to run as from the period cargo is located in the Terminal yeard for shipment. Storage time ends with vessel ETB sent to the users. For purposes of computation of storage free time storage and billable storage, it should be taken in consideration is the so-called 'Billable ETB' that is published in the "Container Ship Programme" option on our website and defined in the Tariffs Regulation and Commercial Policy of APM

Terminals Callao S.A. (1) First time to an a storm the end of complete discharge operation of the vessel or from the moment the container's located in the Terminal yard for shipment. Free time does not apply to full transshipment containers with overrized cargo; that is, for this type of container, the use of operating area is billed from the first day.

(n1) The service service should be used of the service (n15): This price is applicable to all discharged or shipped containers through the Terminal including restow containers (via quay and in the same hold), that require such service. This is applicable for this operation, whether it is connection or disconnection. (n16): That applicable to a discourse (including). (n17): Proce applicable to a store store approximation of the same hold, that require such service. This is applicable for this operation, whether it is connection or disconnection. (n16): This is applicable to a store work. (n16): This is applicable to a store containers via quar, including inspection and monitoring if necessary. This is an one fine payment to be applicable for total energy supply period provided to re-store refer containers. (n16): This is applicable to a store containers via quar, including inspection and monitoring if necessary. The supplicable for total energy supply period provided to re-store refer containers. (n16): This is applicable to network and the store containers via quar, including inspection and monitoring if necessary. The supplicable for total energy or gains the containers (in quark and interview) and interview of quark and in

(n21): Service that involves moving the container between port terminals from the North Terminal and to the South Terminal whenever a container is unloaded via the North Terminal and it is required to be shipped via the South Terminal and/or viceversa.

Section 1.3	Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
1.3.1	Use of Operational Area - Import Full Containers (except Transhipment) (n22)										
1.3.1.1	48 hours to import and/or discharge container (Free time included in Standard Service) (n23)	Regulated				Fre	e				
1.3.1.2	Day: 3 – 6 (Tariff per day)		Per TEU/day			42.40	7.63				
1.3.1.3	Day 7 (Price per day) container 20 and 40 foot		Por TEU			90.10	16.22				
1.3.1.4	Days: 8 – 10 (Price per day)	Non Regulated	Per TEU/day			44.50	8.01				
1.3.1.5	Days: 11 – 20 (Price per day)	Non Negulated	Per TEU/day			44.50	8.01				
1.3.1.6	Days: 21 – 28 (Price per day)		Per TEU/day			44.50	8.01				
1.3.1.7	Day: 29 onwards (Price per day)		Per TEU/day			44.50	8.01				
1.3.2	Use of Operational Area - Export Full Containers (except Transhipment) (n22)										
1.3.2.1	72 hours to export and/or loading container (Free time) (n24)	Regulated				Fre					
1.3.2.2	Days: 4 – 6 (Tariff per day)		Per TEU/day			39.80	7.16				
1.3.2.3	Day 7 (Price per day) container 20 and 40 foot		Por TEU			90.10	16.22				
1.3.2.4	Days: 8 – 10 (Price per day)	Non Regulated	Per TEU/day			44.50	8.01				
1.3.2.5	Days: 11 – 20 (Price per day)	Hon Hogulatoo	Per TEU/day			44.50	8.01				
1.3.2.6	Days: 21 – 28 (Price per day)		Per TEU/day			44.50	8.01				
1.3.2.7	Day: 29 onwards (Price per day)		Per TEU/day			44.50	8.01				

13.27 Day: 23 orwards (Price per day) (27): Trinst and prices from in sub-contrainers that have been discharged/tadded via the North Terminal as per conditors established in the Concession Contract. (23): Trinst and prices from in sub-contract from the day and prices from in sub-contrainers that have been discharged/tadded via the North Terminal as per conditors established in the Concession Contract. (23): Frem to nu as from the end of complete cargo discharge operation are statist from the data of the data with a container is withdrawn from the year. Free time does not apply to full transhipment containers with oversized carge; that is, for this type of container, the use of operating area is billed from the first day. (24): Free time journ as from the end of complete cargo discharge operation area (stranged) and the data with a container is withore the data with oversized carge; that is, for this type of container, the use of operating area is billed from the first day. (24): Free time journ as from the end of complete cargo discharge operation area (stranged) a

Section 1.4	Temporary Depot Special Service - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (I
1.4.1	Temporary Depot Integrated Service for dry import containers (n25)										
1.4.1.1	20' Container - Includes use of area until the 10th day		Per container			302.10	54.38				
1.4.1.2	40' Container - Includes use of area until the 10th day		Per container			435.70	78.43				
1.4.1.3	Use of area during days 11 - 20 (Price per day) (n26)	Non Regulated	Per TEU/day			20.00	3.60				
1.4.1.4	Use of area during days 21 – 28 (Price per day) (n26)		Per TEU/day			25.00	4.50				
1.4.1.5	Use of area from 29th day onwards and other services (n26)			See section 1.3.1.	7 and others from	section 1.5					
1.4.2	Temporary Depot Integrated Service for dry export containers (n27)										
1.4.2.1	20' Container - Includes use of area until the 7th day		Per container			243.10	43.76				
1.4.2.2	40' Container - Includes use of area until the 7th day		Per container			375.70	67.63				
1.4.2.3	Use of area days 8 – 10 (Price per day) (n28)	Non Regulated	Per TEU/day			10.00	1.80				
1.4.2.4	Use of area days 11 – 20 (Price per day) (n28)	Non Rogalated	Per TEU/day			14.00	2.52				
1.4.2.5	Use of area during days 21 – 28 (Price per day) (n28)		Per TEU/day			20.00	3.60				
1.4.2.6	Use of area from 29th day onwards and other services (n28)			See section 1.3.2.	7 and others from	section 1.5					
1.4.3	Temporary Depot integrated service for reefer import containers (n29)										
1.4.3.1	20' Container - Includes use of area until the 7th		Per container			374.60	67.43				
1.4.3.2	40' Container - Includes use of area until the 7th	Non Regulated	Per container			500.90	90.16				
1.4.3.3	Use of area during days 8 – 10 (Price per day) (n30)		Per TEU/day			15.00	2.70				
1.4.3.4	Use of area from day 11 onwards and other services (n30)		See si	ction 1.4.1.3, 1.4.1.4	and 1.3.1.7 and ot	hers from section 1.5					
1.4.4	Temporary Depot integrated service for reefer export containers (n31)										
1.4.4.1	20' Container - Includes use of area until the 7th		Per Container			341.60	61.49				
1.4.4.2	40' Container - Includes use of area until the 7th	Non Regulated	Per Container			448.90	80.80				
1.4.4.3	Use of area from day 8 onwards and other services (n32)		See	section 1.4.2.3 to 1.4	.2.5. 1.3.2.7 and of	hers of section 1.5					

 14.4.3
 Use of area from day 8 onwards and other services (n32)
 See section 1.4.2.3 to 1.4.2.5, 1.3.2.7 and others of section 1.5

 (n25): This all be applicable hencew FAH Terminals Gallands S.A. has been normality as Termorary depot, (034). Service includes Cargo Portion of Standard Service, use of operational area until the 10th day, documentation revision, termorary depot documentation emission (vidarab), customs transmission. Calculation of use of operational area is from the date of complete cargo discharge of the vessel and finahes when the container within the standard service.

 (n26): This all be applicable from the container within the finahes of section 1.4.1.3 is applicable for days 21 to 28, and price of section 1.4.1.1.7 is applicable for days 22 to 28, and price of section 1.4.1.1.7 is applicable for days 22 to 28, and price of section 1.4.1.1.7 is applicable for days 29 to 28. For the other services, the sections 1.5.1, 1.5.3, 1.5.4 and 1.5.5 are applicable independently from the number of storage days.

(p27); This shall be applicable whenever APM Terminals Callao S.A. has been nominated as Temporary Depot (0014). Service induces Cargo Portion of Standard Service, use of operational area until the 7h day, documentation revision, temporary depot documentation emission (volante), austoms transmission. Use of operational area in calculated from the date the container is placed in the container system of Standard Service, use of operational area until the 7h day, documentation revision, temporary depot documentation emission (volante), austoms transmission. Use of operational area in calculated from the date the container is placed in the container system of Standard Service, use of operational area is calculated from the date the container is placed in the Container of That is published in the "Container Ship Programme" option on our website and defined in the Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A. First two days (48 hours) are free of drarge and are included within standard service. [n28]: Price of section 1.4.2.3 is applicable to use of operational area from days 8 to 10 of Temporary Depot, price of section 1.4.2.4 is applicable for days 21 to 28, and price of section 1.4.2.5 is applicable for days 22 to 28. and price of section 1.4.2.5 is applicable for days 29 onwards. For the other services, the sections 1.5.1, 1.5.3, 1.5.4 and 1.5.5, are applicable for days 11 to 28. price of section 1.4.2.5 is applicable for days 29 onwards. For the other services, the sections 1.5.1, 1.5.3, 1.5.4 and 1.5.5, are applicable independently from the number of storage days.

(r/2): This shall be applicable whenever APM Teminals Callao S.A. has been nominated as Temporary Depot (014). Service industes Cargo Portion of Standard Service, use of operational area until the 7th day, energy supply by reefer containers until the 7th day, inspection and monitoring until the 7th day, documentation revision, temporary depot documentation emission (volante), customs transmission. Use of operational area is calculated is from the date of complete cargo discharge of the vessel until the container is withdrawn from the container yard. First two days (48 hours) are free of charge and are included within the standard service.

(n3); Price of section 1.4.3.3 is applicable to use of operational area from days 8 to 10 of Temporary Depot. price of section 1.5.2.4 is applicable for days 21 to 28; and price of section 1.3.17 is applicable for days 29 onwards. Price of section 1.5.2.3 is applicable for days 8 to 10 of Temporary Depot. price of section 1.5.2.4 is applicable for days 21 to 28; and monitoring for days 8 to 10.3.17 is applicable for days 29 onwards. Price of section 1.5.2.3 is applicable for days 8 to 10 of Temporary Depot. price of section 1.5.2.4 is applicable for days 21 to 28; and monitoring for days 8 to 10.3.17 is applicable for days 29 onwards. Price of section 1.5.2.3 is applicable for days 8 to 10 of Temporary Depot. price of section 1.5.2.4 is applicable for days 29 to 28; and monitoring for days 8 to 10.3.17 is applicable for days 29 onwards. Price of section 1.5.1, 1.5.3, 1.5.4 and 1.5.5. are applicable independently from the number of storage days.

(n1): This thal be applicable whenever APM Terminals Callato S A. has been nominated as Temporary Dapet (2014). Service inductes Carep Portion of Standard Service, use of operational area until the 7h day, energy supply to reefer containers until the 7h day, inspection and monitoring until the 7h day, inspection and monitoring until the 7h day, focumentation envision. Iteration envision is the so-called "Bitable ETE" that is published in the Container Synthematic Terminals Callato S A. has been nominated as Temporary depot documentation envision. Iteration envision is the so-called "Bitable ETE" that is published in the Container Synthematic Terminals Callato S A. First two does (48 hours) as the exclusion is the exclusion and resolution in the tartific Regulation and Commercial Policy of APM Terminals Callato S A. First two does (48 hours) as the exclusion as the exclusion and the internation envision is the so-called "Bitable ETE" that is published in the Container Synthematic Terminals Callato S A. First two does (48 hours) as the exclusion and the internation envision is the so-called "Bitable ETE" that is published in the Container Synthematic Terminals Callato S A. First two does (48 hours) are to channer and envictor with the Standard service.

(n/2) Price of section 1.4.2.3 is applicable to use of operational area from days 8 to 10 of Temporary Depot. price of section 1.4.2.4 is applicable for days 21 to 28; and price of section 1.3.2.7 is applicable for days 29 onwards. Price of section 1.5.2.3, is applicable for energy supply for day 80 non-ards and price of section 1.5.2.4 is applicable for days 21 to 28; and price of section 1.3.2.7 is applicable for days 29 onwards. Price of section 1.5.2.3, is applicable for energy supply for day 80 non-ards and price of section 1.5.2.4 is applicable for days 21 to 28; and price of section 1.3.2.7 is applicable for days 2.1 to 28; and price of section 1.3.2.7 is applicable for energy supply for day 80 non-ards and price of section 1.5.2.4 is applicable for days 2.1 to 28; and price of section 1.3.2.7 is applicable for days 2.1 to 28; and price of section 1.3.2.7 is applicable for days 2.1 to 28; and price of section 1.3.2.7 is applicable for days 3.1 to 3.3, 1.5.4 and 1.5.5, are applicable independently from the number of storage days.

Section 1.5	Other Special Services for Containers (Port Terminal & Tempory Depot) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff) Vessel (IGV)	Cargo (Tariff) Cargo (IGV
1.5.1	Other Special Services for Containers (Port Terminal & Tempory Depot) - Services to the Cargo Special Cargo	Nature	onit or charge	vesser (Tariff)	vesser (IOV)	Gargo (Tanin)	Gargo (IGV)	vessel (Tallil) vessel (IGV)	Cargo (Tarili) Cargo (IGV
1.5.1	Special cargo Treatment of Hazardous cargo containers IMO class 1 (n33		Per TEU			270.40	48.67		
1.5.1.1	Treatment of Hazardous cargo containers IMO class 5.2, class 6.2 and class 7 (n3)		Per TEU			270.40	48.67		
1.5.1.2			Per TEU Per TEU			270.40	46.67		
1.5.1.3	Treatment of Hazardous cargo IMO 9 (n33 and n34) Treatment of Hazardous cargo other IMO classes (n33)		Per TEU Per TEU			125.00	22.50		
1.5.1.4		Non Regulated	Container-Move			159.00	28.62		
	Supply of special equipment for handling 20° containers with oversized cargo in the container yard					330.00	59.40		
1.5.1.6	Supply of special equipment for handling 40° containers with oversized cargo in the container yard		Container-Move Per container/day			25.00	59.40 4.50		
	Containers with oversized cargo - use of operational area (n35)						4.50		
1.5.1.8 1.5.2	High Cube Containers 96* (n35) Reefer Container		Container			23.00	4.14		
1.5.2			Des sentsionalises			0.00	0.00		
	Energy - 20 and 40° container (n36 and n37)	Regulated	Per container/hour			0.00	0.00		
1.5.2.2	Inspection and monitoring - 20° and 40° container (n36 and n37)		Per container/day						
1.5.2.3	Energy - 20' and 40' container (Price per day or fraction day) (n38)		Per container/day			78.00	14.04		
1.5.2.4	Inspection and monitoring - 20' and 40' container (n38)	Non Regulated	Per event			28.30	5.09		
1.5.2.5	Pre-cooling empty container (n39)		Per Container			39.70	7.15		
1.5.2.6	Provision of cold antechamber (n40)		Per Unit/day			250.00	45.00		
1.5.2.7	Pre-cooling empty container (n39)	Regulated	Per Container			39.70	7.15		
1.5.2.8	Assembly and disassembly of clip system on reefer Gensets (n36)		Per complete cycle			0.00	0.00		
1.5.3	Cargo Handling and horizontal movement operations								
1.5.3.1	Additional Movement to Standard Service in the terminal as per request of the user or authorities (n36)	Developed	Per Container			0.00	0.00		
1.5.3.2	Cargo inspection with groups o gangs - 20' and 40' container (n36)	Regulated	Per Container			0.00	0.00		
1.5.3.3	Cargo inspection with forklifts - 20' and 40' container (n36)		Per Container			0.00	0.00		
1.5.3.4	Consolidation/deconsolidation of containers with dry cargo (with forklift) (n36)		Per Container			176.00	31.68		
1.5.3.5	Consolidation/deconsolidation of containers with dry cargo (with crew or crew-forklift) (n36)		Per Container			200.00	36.00		
1.5.3.6	Consolidation/deconsolidation of containers with dry cargo (with coil handling forklift or another type of special forklift) (n36)		Per Container			230.00	41.40		
1.5.3.7	Consolidation/deconsolidation of containers with dry cargo (with forklift) including Gate In/Gate Out of empty container (n36)		Per Container			230.00	41.40		
1.5.3.8	Consolidation/deconsolidation of containers with dry cargo (with crew or crew-forklift) including Gate In/Gate Out of empty container (n36)		Per Container			250.00	45.00		
1.5.3.9	Consolidation/deconsolidation of containers with dry cargo (with coil handling forklift or another type of special forklift) including Gate In/Gate Out of empty container (n36)	Non Regulated	Per Container			280.00	50.40		
1.5.3.10	Consolidation/deconsolidation of containers with refrigerated cargo		Per Container			660.00	118.80		
1.5.3.11	Consolidation/deconsolidation of containers with refrigerated cargo including Gate In/Gate Out of empty containers		Per Container			880.00	158.40		
1.5.3.12	Consolidation/deconsolidation of containers with Mobile Harbour Cranes (Price per shift or fraction shift) (n41)		Per shift			1,250.00	225.00		
1.5.3.13	Partial consolidation/deconsolidation of containers with dry cargo		Per Container			100.00	18.00		
1.5.3.14	Depalletizing/palletizing (removal of straps and/or strapping) (n42)		Per Pallet			10.00	1.80		
1.5.4	Other Services								
1.5.4.1	Breakdown as per instructions of freight forward agent (n43)		Per B/L			23.00	4.14		
1.5.4.2	Bill of lading breakdown (n43)		Per breakdown			23.00	4.14		
1.5.4.3	Sweeping of empty container		Per container			6.00	1.08		
1.5.4.4	Simple cleaning of container		Per container			24.00	4.32		
1.5.4.5	Chemical cleaning of container		Per container			48.00	8.64		
1.5.4.6	Container repair	Non Regulated	Per Activity			Per q			
1.5.4.7	Seal placement services		Per seal			12.00	2.16		
1.5.4.8	Additional inspection or revision of seals (n20)		Per seal			12.00	2.16		
1.5.4.9	Labels Placing or Removal Services (n44)		Per container			9.00	1.62		
1.5.4.10	Additional weighing of containers (n45)		Per container			90.00	16.20		
1.5.4.11	Seal breaking and/or container opening without cargo handling (n40)		Per container			40.00	7.20		
1.5.5	Depot for Empty Containers								
1.5.5.1	Gate In (n46)	Non Regulated	Per container			240.00	43.20		
1.5.5.2	Gate Out (n46)		Per container			272.00	48.96		

(33) In the event that a container carries several dasses of IMO cargo, the higher price amount of IMO classes will be applicable as one sole charge. Price of service (subsection 1.5.1.4) does not apply for amonium nitrate IMO class 5.1, unless the user requires the service.

(n34): The price of this service (subsection 1.5.1.3) does not apply for Fishmeal - IMO Class 9 previously declared as such in the Container Announcement List (CAL) for shipping as per procedures

(n35): This service covers additional capacity of containers and or stacks of full containers (except transhipment). In the case of high-scabe containers the price is independant from the number of storage days at the Terminal and this is not applicable for containers of Temporary Depot service (section 1.4 service). In the case of containers with oversized cargo, the use of target service (section 1.4 service) and the section 1.4 service).

(mu); rhospiticable ball users (monogine or representative); mon Arvia: ferminas classes, the selection and or selection and

(n43): It is charged to each breakdown B/L mother, B/L son or B/L grand

(n4): Respective labels shall be placed according to cargo dassification as per IMDG code for export operations. For import operations, placing of label shall take place whenever a discharged container does not have a label or corresponding label, in such case, nervoral of labels will take place. Label removal shall also place when containers are returned for their reutilization or in those cases when these do not correspond to IMDG code (or export operations. For import operations, placing of label shall take place whenever a discharged container does not have a label or corresponding label, in such case, nervoral of labels will take place. Label removal shall also take place when containers are returned for their reutilization or in those cases when these do not correspond to IMDG code, for placing the current label in stand.

(InS) Weighing of containers shall be executed at the facilities of the North Terminal as per request of the user (cargo owner or consignee) or the authorities. The charge of this service is applicable for containers that suffered no modification (from physical inspection or other reason), and which additional weighing result does not vary more than 3% of registered weight from standard service weighting.

(n46): This service shall not be applicable for shipping, unloading or transhipment services for empty containers set forth in Section 1.1.2 and 1.1.3 of current Tariff Schedule.

SECTION 2	В	REAK BULK CARGO S	SERVICES				BREAK BULK CARGO SERVICES											
Section 2.1	Standard Services	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)							
2.1.1	Standard Service to the Vessel - Services to the Vessel																	
2.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.69	0% Rate			1.42	Tasa 0%									
2.1.2	Standard Service to Break Bulk Cargo																	
2.1.2.1	Loading and Unloading of Break Bulk cargo (n47)	Regulated	Per Ton	2.50	0% Rate	15.60	2.81	2.50	0% Rate	15.00	2.70							
(n47): Vessel por	tion of this tariff includes transference of the cargo from vessel to the quay, lashing/unlashing and tally. Payment of Vessel portion will be applicable to the shipping line or cargo consignee, upon c	onditions set forth in the m	aritime shipping contract. If Ve	ssel portion of this tarif	ff is charged to the ca	rgo consignee, then IC	GV (IVA) shall be app	plicable to this tariff.										
Section 2.2	Special Services - Services to the Vessel (n6)	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)							
2.2.1	Hatches or Hold Lids (n7)																	
2.2.1.1	Movement of ISO hatches	Regulated	Per move	0.00	0% Rate													
2.2.2	Re-stow of Break Bulk Cargo																	
2.2.2.1	Re-stow via vessel	Developed	Per ton	19.15	0% Rate			17.00	Tasa 0%									

2.2.2.2	Re-stow via quay (n48)	riogulation	Per ton	38.64	0% Rate			34.00	Tasa 0%		
2.2.3	Transhipment of Break Bulk cargo										
2.2.3.1	Transhipment of Break Bulk cargo - Complete cycle (n49)	Non Regulated	Per ton	25.60	0% Rate						
2.2.4	Special Cargo										
2.2.4.1	Loading/Unloading of project cargo without mobile harbor crane (n50)	Regulado	Por tonelada o m3			35.33	Tasa 0%			31.90	Tasa
2.2.4.2	Loading/Unloading of project cargo with mobile harbor crane (n50)	Negulado	Por tonelada o m3			49.24	Tasa 0%			43.70	Tasa
: This is ap	licable to the cargo that is re-stowed in the same hold (level) or in another hold (level) of the vessel which in both cases cargo requires to be temporarily transferred to the quay.										
): Price of s	vice includes 3 free days of storage which applies after completion of cargo discharge operation of the vessel.										
	licable to all cargo that cannot be transferred by its own means, weighing over 35 tons or being over 50 m3 per unit. The unit of charge shall be applicable to the larger measure between tonnage			ping line or cargo consi	gnee according to th	e maritime shipping cor	ntract. Tariff will be ch	harged to the user wh	o indicates the contr	ract of maritime transpor	t. The tariff
des only on	movement (either unloading or loading); if the project cargo has to be unloaded and then loaded (as happens with full cycle transshipment) entirely at the Terminal, which involves two movements	then the tariff for this serv	ice will be billed twice.								
ection 2.3											
	Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV
2.3.1	Use of Operational Area - All traffic					F					
2.3.1.1	Days: 1 - 3 (Free time - included in Standard Service)	Regulated				Fre					
2.3.1.2	Days: 4 – 10 (Tariff for whole period or fraction period)	New Developed	Per ton			26.00	4.68				
2.3.1.3	Days: 11 - 20 (Tariff per day or fraction day)	Non Regulated	Per ton/day			2.90	0.52				
2.3.1.4	Days: 21 - onwards (Tariff per day or fraction day)		Per ton/day			4.20	0.76				
	Development of Transport Development Operations to the Operation	Mature	Unit of charge	Vessel (Tariff)	Marcal (10)	0 (T	0	Marcal (7.110)	Marcal (10)	Cargo (Tariff)	0 // 824
ection 2.4		Nature	Unit of charge	vessel (Tariff)	vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	vessel (IGV)	Cargo (Tariff)	Cargo (IGV
2.4.1	Temporary Depot Integrated Service - All traffic (n51)					00.00					
2.4.1.1	Break Bulk Cargo - Includes use of operational area until day 10		Per ton			29.00	5.22				
2.4.1.2	Use of operational area from days 11 – 20 (Price per day of day fraction)	New Developed	Per ton/day			1.50	0.27				
2.4.1.3	Use of operational area from day 21 onwards	Non Regulated	Per ton/day			4.20	0.76				
2.4.1.4	Break Bulk Cargo - Includes use of operational area until day 30 (n52)		Per ton			70.00	12.60				
2.4.1.5	Other services (n53) licable whenever APM Terminals Callao S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability). It is				.3.1.4 and 2.5.2 to						
: For the of	tion 24.15 is applicable for use of operational are of temporary deposit of subsection 24.14 from day 31 and onwards. er services, tartifs and prices of sections from 5.2 to 2.5.4 are applicable independently from the number of abrains days of break buck cargo in the Terminal and regardless of the temporary depoint of temporary depo	osit services chosen (eithe Nature		I.4). Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IG
): For the of ection 2.5	er services, latifs and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break but cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo	·····	from section 2.4.1.1 or 2.4. Unit of charge		Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV
): For the of	er services, tariffs and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep	Nature			Vessel (IGV)	Cargo (Tariff) 11.00	Cargo (IGV) 1.98	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV
: For the of ection 2.5 2.5.1 2.5.1.1	er services, tailfs and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Other: Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All traffic (r/S4)	·····	Unit of charge		Vessel (IGV)			Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV
): For the of ection 2.5 2.5.1 2.5.1.1 2.5.2	er services, tarlis and prices of sections from 2.52 to 2.5.4 are applicable independently from the number of storage days of break but rango in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All traffer (ris5) Days, 1 - 30 (Price per month or fraction month)	Nature	Unit of charge		Vessel (IGV)			Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff) 31.90	
Example: For the of 2.5.1 2.5.1.1 2.5.2 2.5.2.1	er services, tarifs and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break buk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All traffic (n54) Days 1 - 3.0 (Price per month or fraction month) Special Cargo	Nature Non Regulated	Unit of charge Per m2		Vessel (IGV)	11.00	1.98	Vessel (Tariff)	Vessel (IGV)		
Ection 2.5 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2.1 2.5.2.2	er services, tarlis and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All Itaffic (ris5) Days: 1 - 30 (Price per month or fraction month) Secial Cargo LadingUnbacking of project cargo without mobile harbor crane (ris5)	Nature	Unit of charge Per m2 Per ton or m3		Vessel (IGV)	11.00	1.98	Vessel (Tariff)	Vessel (IGV)	31.90	
Example: For the of 2.5.1 2.5.1 2.5.2 2.5.2.1 2.5.2.2 2.5.2.2 2.5.2.3	er services, tarifs and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break buk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Days: 1 - 30 (Price per month or fraction month) Special Cargo of project cargo without mobile harbor crane (n50) Laading Unbedding of project cargo with mobile harbor crane (n50)	Nature Non Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3		Vessel (IGV)	11.00 35.33 49.24	1.98 6.36 8.86	Vessel (Tariff)	Vessel (IGV)	31.90	
Example: For the of 2.5.1 2.5.1.1 2.5.2 2.5.2.1 2.5.2.2 2.5.2.3 2.5.2.4	er services, tarlis and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All tarlife (ris5) Days: 1 - 30 (Price per month or fraction month) Special Cargo Laading Unbadding of project cargo without mobile harbor crane (ris5) Laading Unbadding of project cargo with nobile harbor crane (ris5) Laading Unbadding of project cargo with nobile harbor crane (ris5)	Nature Non Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton		Vessei (IGV)	11.00 35.33 49.24 20.10	1.98 6.36 8.86 3.62	Vessel (Tariff)	Vessei (IGV)	31.90	
ection 2.5 2.5.1 2.5.2 2.5.2.1 2.5.2.2 2.5.2.1 2.5.2.2 2.5.2.3 2.5.2.4 2.5.3	er services, terifs and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot. All traffic (n54) Days: 1-3.0 (Price per month or fraction month) Special Cargo Leading Unicoding of project cargo without mobile harbor crane (n50) Leading Unicoding of project cargo with mobile harbor crane (n50) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - Indirect treatment (n55)	Nature Non Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton		Vessei (IGV)	11.00 35.33 49.24 20.10	1.98 6.36 8.86 3.62	Vessel (Tariff)	Vessei (IGV)	31.90	
E For the of 2.5.1 2.5.1.1 2.5.2 2.5.2.1 2.5.2.2 2.5.2.3 2.5.2.4 2.5.3 2.5.3.1	er services, tarlis and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Use of Simple Depot - All tarlife (ris5) Use of Simple Depot - All tarlife (ris5) Depst - 1-30 (Price per month of traction month) Seceial Careo Laading Unbacking of project cargo without mobile harbor crane (ris5) Laading Unbacking of project cargo with nobile harbor crane (ris5) Special tractment to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Special tractment to hazardous cargo using equipment and additional personnel - indirect treatment (ris5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (ris5) Cargo Handling and horizontal movement operations	Nature Non Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton		Vessel (IGV)	11.00 35.33 49.24 20.10 24.10	1.98 6.36 8.86 3.62 4.34	Vessel (Tariff)	Vessel (IGV)	31.90	
Example: For the off 2.5.1.1 2.5.2.1 2.5.2.1 2.5.2.2 2.5.2.3 2.5.2.4 2.5.3.1 2.5.3.1 2.5.3.2	er services, terlis and prices of sections from 2.52 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All traffic (ris5) Days: 1-30 (Pitos per month or faction month) Special Cargo Loading/Unoading of project cargo without mobile harbor crane (ris5) Loading/Unoading of project cargo without mobile harbor crane (ris5) Special Instantion to August of sections and additional personnel - direct treatment (ris5) Special Instantion to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Cargo Handling and horizontal movement operations Additional Movement as per requised to the user or complement authority Handling for registry of break bulk cargo (ris6) Deconsolidation Corenciation of the user cargo with MAFIs (ris7)	Nature Non Regulated Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton Per ton		Vessel (IGV)	11.00 35.33 49.24 20.10 24.10 4.00	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58	Vessel (Tariff)	Vessel (IGV)	31.90	
Evention 2.5 2.5.1 2.5.1.1 2.5.2 2.5.2.1 2.5.2.2 2.5.2.3 2.5.2.4 2.5.3.1 2.5.3.2 2.5.3.3 2.5.3.4	er services, tarth and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Date of Simple Depot - All tartifie (nS5) Use of Simple Depot - All tartifie (nS5) Dates (1 - 20) (Price per month or fraction month) Special Cargo Laading/Unbading of project cargo without mobile harbor crane (nS0) Laading/Unbading of project cargo with nobile harbor crane (nS0) Special trastment to hazardous cargo using equipment and additional personnel - direct treatment (nS5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (nS5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (nS5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (nS5) Additional Movement as per request of the user or completent authority Handing for registry of brack bulk cargo (nK6) Deconsolitation/Consolitation of Brack Bulk cargo in MRF1s (nS7) Deconsolitation of Special bulk cargo (nK6)	Nature Non Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per Shift	Vessel (Tariff)		11.00 35.33 49.24 20.10 24.10 4.00 4.00	1.98 6.36 8.86 3.62 4.34 0.72 0.72	Vessel (Tariff)	Vessei (IGV)	31.90	Ę
E For the of 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.5 2.5.5 2.5.5 2.5.5 2.5.5 2.5.5 2.5.5 2.5.5 2.	er services, terlifs and prices of sections from 2.52 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All traffic (ris5) Days: 1-30 (Pitoe per month or Taction month) Special Cargo LaadingUnbacing of project cargo without mobile harbor crane (ris5) LaadingUnbacing of project cargo without mobile harbor crane (ris5) Special Instammer to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Special Instammer to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Cargo Handling and horizontal movement operations Additional Movement as per requised of the user or competent authority Handling for registry of break buik cargo (ris6) Deconsolidation/Consolidation of Break Buik cargo in MAPIs (ris7) Deconsolidation/Consolidation of Break Buik cargo in MAPIs with Mobile Harbour Cranes (Price per shift or fraction shift) (ris6) Deconsolidation/Consolidation of Break Buik cargo in MAPIs with Mobile Harbour Cranes (Price per shift or fraction shift) (ris6)	Nature Non Regulated Regulated	Unit of charge Per m2 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per MAFI		Vessel (IGV)	11.00 35.33 49.24 20.10 24.10 4.00 4.00 281.00 1,250.00	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58 225.00	Vessei (Tariff)	Vessel (IGV)	31.90	
For the of ction 2.5 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.5 2.5.5 2.5.5 2.5.5 2.5.5 2.5.5 2.5.5	er services, tarth and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Date of Simple Depot - All tartifie (nS5) Use of Simple Depot - All tartifie (nS5) Dates (1 - 20) (Price per month or fraction month) Special Cargo Laading/Unbading of project cargo without mobile harbor crane (nS0) Laading/Unbading of project cargo with nobile harbor crane (nS0) Special trastment to hazardous cargo using equipment and additional personnel - direct treatment (nS5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (nS5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (nS5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (nS5) Additional Movement as per request of the user or completent authority Handing for registry of brack bulk cargo (nK6) Deconsolitation/Consolitation of Brack Bulk cargo in MRF1s (nS7) Deconsolitation of Special bulk cargo (nK6)	Nature Non Regulated Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per Shift	Vessel (Tariff)		11.00 35.33 49.24 20.10 24.10 4.00 4.00 281.00	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58	Vessel (Tariff)	Vessel (IGV)	31.90	
E For the of 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.	er services, terlifs and prices of sections from 2.52 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All traffic (ris5) Days: 1-30 (Pitoe per month or Taction month) Special Cargo LaadingUnbacing of project cargo without mobile harbor crane (ris5) LaadingUnbacing of project cargo without mobile harbor crane (ris5) Special Instammer to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Special Instammer to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Cargo Handling and horizontal movement operations Additional Movement as per requised of the user or competent authority Handling for registry of break buik cargo (ris6) Deconsolidation/Consolidation of Break Buik cargo in MAPIs (ris7) Deconsolidation/Consolidation of Break Buik cargo in MAPIs with Mobile Harbour Cranes (Price per shift or fraction shift) (ris6) Deconsolidation/Consolidation of Break Buik cargo in MAPIs with Mobile Harbour Cranes (Price per shift or fraction shift) (ris6)	Nature Non Regulated Regulated	Unit of charge Per m2 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per MAFI	Vessel (Tariff)		11.00 35.33 49.24 20.10 24.10 4.00 4.00 281.00 1,250.00	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58 225.00 3.60	Vessel (Tariff)	Vessel (IGV)	31.90	Ę
): For the of 2.5.1 2.5.1.1 2.5.2.1 2.5.2.1 2.5.2.2 2.5.2.3 2.5.2.4	er services, tarth and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Date of Simple Depot - All tartific (n53) Use of Simple Depot - All tartific (n53) Dates 1 - 30 (Price per month or fraction month) Special Carroo Laading/Unbading of project cargo without mobile harbor crane (n50) Laading/Unbading of project cargo without mobile harbor crane (n50) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Cargo Handling and horizontal movement operations Additional Movement as per request of the user or completent authontly Handing for registry of brak buik cargo (n45) Deconsolitation (Consolitation of Break Buik Cargo in MFIs (n57) Deconsolitation of Marke Buik cargo in MFIs (n59) Deconsolitation of Marke Buik cargo in MFIs (n59) Deconsolitation of Marke (n50) Deconsolitation (Marke (n50)	Nature Non Regulated Regulated	Unit of charge Per m2 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per MAFI	Vessel (Tariff)		11.00 35.33 49.24 20.10 24.10 4.00 4.00 281.00 1,250.00	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58 225.00	Vessel (Tariff)	Vessel (IGV)	31.90	Cargo (IGV £ 7
E For the of ction 2.5 2.5.1 2.5.2.1 2.5.2.1 2.5.2.2 2.5.2.3 2.5.2.4 2.5.3.1 2.5.3.2 2.5.3.2 2.5.3.4 2.5.3.5 2.5.3.6 2.5.4 2.5.4.1 : This servi	er services, terifs and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All traffer (ris5) Days: 1-30 (Pites per month or Tacinon month) Special Cargo Laading Unbading of project cargo with mobile harbor crane (ris5) Laading Unbading of project cargo with mobile harbor crane (ris5) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (ris5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (ris5) Cargo Handling and horizontal movement operations Additional Movement as per request of the user or competent authority Handling for registry of break buk cargo (ris6) Deconsolidation/Consolidation of Have Buk cargo (ris6) Deconsolidation/Cons	Nature Non Regulated Regulated Non Regulated Non Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per Shift Per MAFI Per ton Per ton	Vessel (Tariff) 83.00	14.94	11.00 35.33 49.24 20.10 24.10 24.10 4.00 281.00 1.250.00 20.00 21.00	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58 225.00 3.60 3.60 3.78			31.90 43.70	
E For the of ction 2.5 2.5.1 2.5.2.1 2.5.2.1 2.5.2.2 2.5.2.3 2.5.2.4 2.5.3.1 2.5.3.2 2.5.3.2 2.5.3.4 2.5.3.5 2.5.3.6 2.5.4 2.5.4.1 : This servi	er services, tarth and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Use of Simple Depot - All tartiffer (hSI) Use of Simple Depot - All tartiffer (hSI) Days 1 - 30 (Price per month or fraction month) Special Cargo Laading/Unideding of project cargo with nuclie harbor crane (hSI) Laading/Unideding of project cargo with mobile harbor crane (hSI) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (hSS) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (hSS) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (hSS) Special treatment to hazardous cargo (hSI) Additional Movement as per request of the user or competent authority Handing for registry of break buik cargo (hSI) Deconsolitation/Consolitation of Break Buik argo (hSI) Deconsolitation/Consolitation of MARE (hGP) Beconsolitation/Consolitation of MARE (hGB) Beconsolitation/Consolitation of MARE (hGB)	Nature Non Regulated Regulated Non Regulated Non Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per Shift Per MAFI Per ton Per ton	Vessel (Tariff) 83.00	14.94	11.00 35.33 49.24 20.10 24.10 24.10 4.00 281.00 1.250.00 20.00 21.00	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58 225.00 3.60 3.60 3.78			31.90 43.70	
Ction 2.5 2.5.1 2.5.1 2.5.2 2.5.2.1 2.5.2.2 2.5.2.4 2.5.2.3 2.5.2.3 2.5.3.1 2.5.3.3 2.5.3.4 2.5.3.4 2.5.3.5 2.5.3.6 2.5.4 2.5.4 2.5.4.1 2.5.4.1	er services, tarth and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Use of Simple Depot - All traffic (n54) Use of Simple Depot - All traffic (n54) Depot - All traffic (n55) Special transmitts of braizardos cargo using explained and additional personnel - direct treatment (n55) Special transmitts of hourandos cargo using explained and additional personnel - direct treatment (n55) Special treatment to hexandos cargo using explained and additional personnel - direct treatment (n55) Special treatment to hexandos cargo using explained and additional personnel - indirect treatment (n55) Cargo Handling and hourointal movement operations Additional Wowment as per request of the user or completent authoffy Handing for registry of brank buk cargo (n56) Deconsolitation/Consolitation of Brank Buk cargo (n56) Deconsolitation of MARIs (n59) Deconsolitation of MARIs (n59) Deconsolitation of MARIs (n59) Deconsolitation of MARIs (n59) Deconsolitation of MARIS (n50) Determined and provide and avoid basel (80) Other services Additional weeking (n16) idea ent include extra movement) refer this moves on a splonde as Simple Deconsolitation of break Buk cargo in MARIS (n57) Deconsolitation Deconsolitation of Deconsoli	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per Shift Per MAFI Per ton Per ton	Vessel (Tariff) 83.00 rgo (n containers main	14.94 ly) or for another car	11.00 36.33 49.24 20.10 24.10 4.00 28.100 1.250.00 20.00 21.00 30 that has been unloa	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58 225.00 3.60 3.78 ded in container and	is then stored as Bre	ek Bulk. Standard se	31.90 43.70	Jed to Break B
Ction 2.5 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.4 2.5.4 2.5.4 2.5.4 2.5.4 2.5.4 2.5.4 2.5.4 2.5.2 2.5.3 2.5.5.3 2.5.5.3 2.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5	er services, terifs and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dep Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All traffer (ris5) Days: 1-30 (Pites per month or Tacinon month) Special Cargo Laading Unbading of project cargo with mobile harbor crane (ris5) Laading Unbading of project cargo with mobile harbor crane (ris5) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (ris5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (ris5) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (ris5) Cargo Handling and horizontal movement operations Additional Movement as per request of the user or competent authority Handling for registry of break buk cargo (ris6) Deconsolidation/Consolidation of Have Buk cargo (ris6) Deconsolidation/Cons	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated to the user. Tarific of se	Unit of charge Per ton or m3 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per ton Per MAFI Per ton Per tuck hipped as another type of c atom 2.5.2.3 and 2.5.2.4 inc	Vessel (Tariff) 83.00 rgo (n containers main	14.94 ly) or for another car	11.00 36.33 49.24 20.10 24.10 4.00 28.100 1.250.00 20.00 21.00 30 that has been unloa	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58 225.00 3.60 3.78 ded in container and	is then stored as Bre	ek Bulk. Standard se	31.90 43.70	Jed to Break B
Ction 2.5 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.4 2.5.4 2.5.4 2.5.4 2.5.4 2.5.4 2.5.4 2.5.3 2.5.4 2.5.5.4 2.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5	er services, tarifs and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dapo Use of Simple Depot - All traffic (n54) Use of Simple Depot - All traffic (n54) Use of Simple Depot - All traffic (n54) Decomposition of project cargo without mobile harbor cares (n55) Ladaring Unicading of project cargo without mobile harbor cares (n55) Ladaring Unicading of project cargo without mobile harbor cares (n55) Special transmitter In hazardous cargo using equipment and additional personnel – direct treatment (n55) Special transmitter In hazardous cargo using equipment and additional personnel – direct treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel – direct treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel – funct treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel – direct treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel – funct treatment (n55) Decomposition of break buk cargo (n50) Decomposition of break buk cargo (n50) Decomposition of break buk cargo (n50) Decomposition of break buk cargo (n50) Determinel (n55) Decomposition of break buk cargo (n50) Determinel (n56) Decomposition of break buk cargo (n50) Decomposition of break buk cargo (n	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated to the user. Tarific of se	Unit of charge Per ton or m3 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per ton Per MAFI Per ton Per tuck hipped as another type of c atom 2.5.2.3 and 2.5.2.4 inc	Vessel (Tariff) 83.00 rgo (n containers main	14.94 ly) or for another car	11.00 36.33 49.24 20.10 24.10 4.00 28.100 1.250.00 20.00 21.00 30 that has been unloa	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58 225.00 3.60 3.78 ded in container and	is then stored as Bre	ek Bulk. Standard se	31.90 43.70	Jed to Break B
For the ol ction 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.3 2.5.4 2.5.4 2.5.3 2.5.4	er services, terls and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dap Other Signel Depot - All Itaffie (ris5) Use of Simple Depot - All Itaffie (ris5) Deposition of the section month) Special Tearment on the tractions month) Special Tearment on the tractions cargo with nucleie harbor crane (ris5) Laading Unbacking of project cargo without mobile harbor crane (ris5) Laading Unbacking of project cargo without mobile harbor crane (ris5) Special Tractinem to hexardous cargo using equipment and additional personnel - direct treatment (ris5) Special Treatment to hexardous cargo using equipment and additional personnel - indirect treatment (ris5) Special Treatment to hexardous cargo using equipment and additional personnel - indirect treatment (ris5) Cargo Handling and horizontal movement operations Additional Movement as per request of the user or completent authontly Hearding for registry of brack buik cargo (ris6) Deconsolitation/Consolitation of Depark Buik cargo in MAFIs (ris7) Deconsolitation of Consolitation of Mark Buik cargo in MAFIs (ris7) Deconsolitation of Deconsolitation of Buik Buik cargo in MAFIs (ris7) Deconsolitation of Deconsolitation of Buik Buik cargo in MAFIs (ris7) Deconsolitation of Deconsolitation of Buik Buik cargo in MAFIs (ris7) Deconsolitation of Deconsolitation of Buik Buik cargo in MAFIs (ris7) Deconsolitation of Deconsolitation of Buik Buik cargo in MAFIs (ris7) Deconsolitation of Deconsolitation of Buik Buik Cargo (ris6) Deconsolitation of Buik Bui	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated to the user. Tarify of see d to the user at the store	Unit of charge Per ton or m3 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per ton Per MAFI Per ton Per tuck hipped as another type of c atom 2.5.2.3 and 2.5.2.4 inc	Vessel (Tariff) 83.00 rgo (n containers main	14.94 ly) or for another car	11.00 36.33 49.24 20.10 24.10 4.00 28.100 1.250.00 20.00 21.00 30 that has been unloa	1.98 6.36 8.86 3.62 4.34 0.72 0.72 50.58 225.00 3.60 3.78 ded in container and	is then stored as Bre	ek Bulk. Standard se	31.90 43.70	Jed to Break B
For the ol ction 2.5 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.4 2.5.5.4 2.5.5.5.5 2.5.5.5.5.5.5.5.5.5.5.5.5.5.5	er services, turth and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dapo Use of Simple Depot - All traffic (n54) Use of Simple Depot - All traffic (n54) Depot - All traffic (n55) Special transmitt in the lardbox cargo using equipment and additional personnel - direct treatment (n55) Special transmitt in the lardbox cargo using equipment and additional personnel - direct treatment (n55) Special transmitt in the anabox cargo using equipment and additional personnel - direct treatment (n55) Depote half (n55)	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated Non Regulated to the user. Tariffo dia to the user. Tariffo dia to the user at the store to the cargo consignee.	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per MAFI Per MAFI Per ton Per ton P	Vessel (Tariff) 83.00 rgo (n containers main use Standard Service to	14.94 ly) or for another car o break bulk cargo. I	11.00 35.33 49.24 20.10 24.10 24.10 4.00 28.100 1,250.00 20.00 21.00 20 that has been unlead 21.00 20 that has been unlead	1.98 6.36 8.86 6.36 2.4.34 0.72 0.72 5.0.58 225.00 3.60 3.60 3.78 ded in container and and personnel are no	is then stored as Bre	ak Bulk. Standard service t	31.90 43.70 ervice shall not be provid ariff to break bulk cargo	ded to Break B
For the ol 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.4 2.5.5.5 2.5.4 2.5.5.5 5.5.5.5 5.5.5.5.	er services, tarth and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dap Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All taffet (r55) Days 1 - 30 (Price per month or fraction month) Special Cargo Laading Unbacing of project cargo with nuclie harbor crane (r50) Laading Unbacing of project cargo with nuclie harbor crane (r50) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (r55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (r55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (r55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (r55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (r55) Decomplications of break buik cargo (r56) Decomplications (Consolidation of Decomplications (r57) Decomplications (r50) Decomplications (r50) Decomplications (r50) Determined (r50) Decomplications (r50) Determined (r50) De	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated Non Regulated to the user. Tariffo dia to the user. Tariffo dia to the user at the store to the cargo consignee.	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per MAFI Per MAFI Per ton Per ton P	Vessel (Tariff) 83.00 rgo (n containers main use Standard Service to	14.94 ly) or for another car o break bulk cargo. I	11.00 35.33 49.24 20.10 24.10 24.10 4.00 28.100 1,250.00 20.00 21.00 20 that has been unlead 21.00 20 that has been unlead	1.98 6.36 8.86 6.36 2.4.34 0.72 0.72 5.0.58 225.00 3.60 3.60 3.78 ded in container and and personnel are no	is then stored as Bre	ak Bulk. Standard service t	31.90 43.70 ervice shall not be provid ariff to break bulk cargo	ded to Break B
For the ol ction 2.5.1 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.4 2.5.4 2.5.4 2.5.3 2.5.3 2.5.3 2.5.4 2.5.4 2.5.4 2.5.4 2.5.4 2.5.3 2.5.3 2.5.3 2.5.4 2.5.4 2.5.4 2.5.4 2.5.3 2.5.3 2.5.4 2.5.4 2.5.4 2.5.4 2.5.5 2.5.5 2.5.4 2.5.4 2.5.4 2.5.3 2.5.4 2.5.	er services, turth and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary daps Use of Simple Depot - All traffic (nA) Use of Simple Depot	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated Non Regulated to the user. Tariffo dia to the user. Tariffo dia to the user at the store to the cargo consignee.	Unit of charge Per m2 Per ton or m3 Per ton or m3 Per ton Per ton Per ton Per ton Per MAFI Per MAFI Per MAFI Per ton Per ton P	Vessel (Tariff) 83.00 rgo (n containers main rude Standard Service to	14.94 ly) or for another car o break bulk cargo. I	11.00 35.33 49.24 20.10 24.10 24.10 4.00 28.100 1,250.00 20.00 21.00 20 that has been unlead 21.00 20 that has been unlead	1.98 6.36 8.86 6.36 2.4.34 0.72 0.72 5.0.58 225.00 3.60 3.60 3.78 ded in container and and personnel are no	is then stored as Bre	ak Bulk. Standard service t	31.90 43.70 ervice shall not be provid ariff to break bulk cargo	ded to Break Bu
For the ol 2.5.7 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.4 1. This servi in this cas 3. 2.5.4 1. This refe Price is a J Service in This is a J This is a J 2.5.4	er services, tarth and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dap Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All taffer (n55) Days 1 - 30 (Price per month or fraction month) Special Cargo Ladding Unbading of project cargo with nuclie harbor crane (n50) Ladding Unbading of project cargo with nuclie harbor crane (n50) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Decomptibilitation (Consolidation of Break Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of MAFIs (n59) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n59) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n59) Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation)) and additional velocity gene to the additional personal of personal as the the period apprint and personal as the the period apprint and personal as the the period apprint and personal as the the period and personal as the the period apprint and pers	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated Non Regulated to cargo that will be finally: at of the user. Terrifly of see de to the user at the store to the cargo consignee. reak bulk cargo from mete	Unit of charge Per m2 Per ton cm3 Per ton cm3 Per ton Per ton Per ton Per ton Per ton Per MAFI Per Shift Per Shift Per MAFI Per ton Per ton Per ton Strome charge of c constraints of the shift of the shift of the shift of the per ton Per ton charge of the shift of t	Vessel (Tariff) 83.00 rge (n containes main tude Standard Service II MAFI taaler). Price II in	14.94 ly) or for another car to break bulk cargo. I voiced to the Cargo	11.00 35.33 49.24 20.10 24.10 4.00 4.00 1.250.00 21.00 20.00 21.00 20.00 21.00 20.00 21.00 20.00	1.98 6.36 8.65 3.62 7.2 50.58 225.00 3.60 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.72 5.25.00 3.72 3.72 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 5.00	is then stored as Bre	ak Bulk. Standard service t	31.90 43.70 ervice shall not be provid ariff to break bulk cargo	ded to Break B
For the ol 2.5 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.2 2.5.3 2.5.4 1. This servi in this cass 3.8 2.5.4 1. This refe Price is a Service in This is ag	er services, turth and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary daps Use of Simple Depot - All traffic (nA) Use of Simple Depot	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated Non Regulated to cargo that will be finally: at of the user. Terrifly of see de to the user at the store to the cargo consignee. reak bulk cargo from mete	Unit of charge Per m2 Per ton cm3 Per ton cm3 Per ton Per ton Per ton Per ton Per ton Per MAFI Per Shift Per Shift Per MAFI Per ton Per ton Per ton Strome charge of c constraints of the shift of the shift of the shift of the per ton Per ton charge of the shift of t	Vessel (Tariff) 83.00 rge (n containes main tude Standard Service II MAFI taaler). Price II in	14.94 ly) or for another car to break bulk cargo. I voiced to the Cargo	11.00 35.33 49.24 20.10 24.10 4.00 4.00 1.250.00 21.00 20.00 21.00 20.00 21.00 20.00 21.00 20.00	1.98 6.36 8.65 3.62 7.2 50.58 225.00 3.60 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.72 5.25.00 3.72 3.72 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 5.00	is then stored as Bre	ak Bulk. Standard service t	31.90 43.70 ervice shall not be provid ariff to break bulk cargo	fed to Break B
For the ol 2.5 2.5.1 2.5.1 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.2 2.5.3 2.5.2 2.5.3 2.5.4 1. This servi in this cass 3.8 2.5.4 1. This refe Price is a Service in This is ag	er services, tarth and prices of sections from 2.52 to 2.54 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary dap Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo Use of Simple Depot - All taffer (n55) Days 1 - 30 (Price per month or fraction month) Special Cargo Ladding Unbading of project cargo with nuclie harbor crane (n50) Ladding Unbading of project cargo with nuclie harbor crane (n50) Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55) Decomptibilitation (Consolidation of Break Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of MAFIs (n59) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n57) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n59) Decomptibilitation (Consolidation of Mark Buik cargo in MAFIs (n59) Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation (Decomptibilitation)) and additional velocity gene to the additional personal of personal as the the period apprint and personal as the the period apprint and personal as the the period apprint and personal as the the period and personal as the the period apprint and pers	Nature Non Regulated Regulated Non Regulated Non Regulated Non Regulated Non Regulated to cargo that will be finally: at of the user. Terrifly of see de to the user at the store to the cargo consignee. reak bulk cargo from mete	Unit of charge Per m2 Per ton cm3 Per ton cm3 Per ton Per ton Per ton Per ton Per ton Per MAFI Per Shift Per Shift Per MAFI Per ton Per ton Per ton Strome charge of c constraints of the shift of the shift of the shift of the per ton Per ton charge of the shift of t	Vessel (Tariff) 83.00 rge (n containes main tude Standard Service II MAFI taaler). Price II in	14.94 ly) or for another car to break bulk cargo. I voiced to the Cargo	11.00 35.33 49.24 20.10 24.10 4.00 4.00 1.250.00 21.00 20.00 21.00 20.00 21.00 20.00 21.00 20.00	1.98 6.36 8.65 3.62 7.2 50.58 225.00 3.60 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.78 4.25.00 3.72 5.25.00 3.72 3.72 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 3.75 5.25.00 5.00	is then stored as Bre	ak Bulk. Standard service t	31.90 43.70 ervice shall not be provid ariff to break bulk cargo	ded to Breat

0201101110											
Section 3.1	Standard Service	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.1.1	Standard Service to the Vessel - Services to the Vessel										
3.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.69	0% Rate			1.42	Tasa 0%		
3.1.2	Standard Service to Ro-Ro Cargo										
3.1.2.1	Loading or Unloading of Ro-Ro Cargo (n61)	Regulated	Per ton	3.50	0% Rate	44.10	7.94	4 3.50	0% Rate	42.80	7.70
(n61): Vessel port	tion of this tariff will be charged to the Shipping line (or representative), unless maritime shipping contract indicates otherwise. If Vessel portion of this tariff is charged to the cargo consignee, then li	GV (IVA) shall be applicab	le to this tariff.								
Section 3.2	Special Services - Services to the Vessel	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.2.1	Re-stow of Ro-Ro cargo										
3.2.1.1	Re-stow via vessel		Per ton	57.92	0% Rate			50.30	Tasa 0%		
3.2.1.2	Re-stow via guav (n62)	Regulated	Per ton	112.88	0% Rate			99.00	Tasa 0%		
3.2.2	Transhipment of Ro-Ro caroo										
3.2.2.1	Transhipment of Ro-Ro cargo - Complete cycle (n63)	Non Regulated	Per ton	74.00	0% Rate						
3.2.3	Use of Operational Area - Transhipment (n64)										
3.2.3.1	Dias: 1 -3 (Tiempo libre - included in the special service ro-ro caroo transhipment)			Fr	ee						
3.2.3.2	Days: 4 – 10 (Price for total period or fraction period)	No Regulado	Per ton	48.80	Tasa 0%						
3.2.3.3	Days: 11 - onwards (Price per day or fraction day)		Per ton/day	20.00	Tasa 0%						
(n62): This is app	plicable to the cargo that is re-stowed in the same hold (level) or in another hold (level) of the vessel which in both cases cargo requires to be temporarily transferred to the quay.										
(n63): Price of ser	rvice includes 3 free days of storage which applies after completion of cargo discharge operation of the vessel.										
(n64): Prices of th	his section are applicable to all types of vehicle and machinery.										
Section 3.3	Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.3.1	Use of Operational Area - All traffic (except transhipment) (n64)										
3.3.1.1	Davs: 1 - 3 (Free time - included in Standard Service)	Regulated				Fr	ee				
3.3.1.2	Davs: 4 - 10 (Price for total period or fraction period)		Per ton			90.00	16.20				
3.3.1.3	Days: 11 - onwards (Price per day or fraction day)	Non Regulated	Per ton/day			20.00	3.60				
Section 3.4	Special Service of Temporary Depot - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.4.1	Temporary Depot Integrated Service - All traffic (n65)										
3.4.1.1	Ro-Ro cargo in general - Includes use of operational area until day 10		Per ton			120.00	21.60				
3.4.1.2	Use of operational area from day 11 onwards (Price per day or fraction day)	Non Regulated	Per ton/day			10.00	1.80				
3.4.1.3	Other services (n66)			S	ee section 3.5						
	licable whenever APM Terminals Callao S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability). It is	ndudes the Standard Sen	rice cargo portion, use of ope	ational area until day '	10, temporary depot	documentation emission	on (volante) and tran	smission to Customs.	First three calendar of	ays are free of charge	and are part of the
Standard Service	. Prices of this section are applicable to all types of vehicles and machinery.										
(n66): Prices of se	action 3.5 apply to other services of Temporary Depot, independently from the number of storage days of RoRo cargo.										
Section 3.5	Other Special Services to Ro-Ro Cargo (Port Terminal & Temporary Depot) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.5.1	Handling Services in Storage area										
3.5.1.1	Handling for registry of Ro-Ro cargo (n67)	New Developed	Per ton			11.00	1.98	3			
3.5.1.2	Extra movement of Ro-Ro cargo (n68)	Non Regulated	Per ton			2.50	0.45	5			
3.5.2	Deconsolidation/Consolidation										

 3.2
 Deconsultation Consolidation
 L
 C

 3.5.2.1
 Desconsidiation Consolidation
 Desconsidiation Consolidation Consolidation
 Desconsidiation Consolidation Consolidation
 Desconsidiation Consolidation Consolidation
 Desconsidiation Consolidation
 Desconsidiation Consolidation
 Desconsidiation Consolidation Consolidation Consolidation Consolidation
 Desconsidiation Consolidation Consolidation Consolidation Consolidation Consolidation Consolidation<

SECTION 4		SOLID BULK CAR	GO								
Section 4.1	Standard Service	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
4.1.1	Standard Service to the Vessel - Services to the Vessel										
4.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.69	0% Rate			1.42	Tasa 0%		
4.1.2	Standard Service to Solid Bulk Cargo - Services to the Cargo										
4.1.2.1	Loading or Unloading of Solid Bulk	Regulated	Per ton			9.37	1.69			8.30	1.49
Section 4.2	Special Services - Services to the Vessel	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
4.2.1	Hatches or Hold Lids(n6)										
4.2.1.1	Movement of ISO hatches	Regulated	Per move	0.00	0% Rate						
Section 4.3	Special Services - Services to the Cargo		Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
4.3.1	Remove/place of artificial separation in vessel's hold										
4.3.1.1	Remove of artificial separation in vessel's hold	Regulated	Per operation			2,000.00	360.00				
4.3.2	Use of Operational Area - All traffic										
4.3.2.1	Days: 1 - 5 (Free time - included in Standard Service) (n71)	Regulated				Fre					
	Days: 6 – 15 (Price for whole period or fraction period)		Per ton			2.90	0.52				
4.3.2.3	Days: 16 – 30 (Price for whole period or fraction period)	Non Regulated	Per ton			4.00	0.72				
4.3.2.4	Days: 31 - onwards (Price per day or fraction day)		Per ton/day			0.40	0.07				
	Other services										
4.3.3.1	Additional weighing (Extra movement is not included) (n72)	Non Regulated	Per truck			21.00	3.78				

(n71): Free days are given to goods which use absorbing towers and allos. (n72): This service shall be provided within facilities of Multipurpose North Terminal as per request of user or authorities. (Extra) movement of sold bulk cargo is requested from its stacking area to the Gates area. Such service does not include extra movement service.

SECTION 5		LIQUID BULK CA	RGO								
Section 5.1	Standard Service	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
5.1.1	Standard Service to the Vessel - Services to the Vessel										
5.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.69	0% Rate			1.42	Tasa 0%		
5.1.2	Standard Service to Liquid Bulk Cargo - Services to the Cargo										
5.1.2.1	Loading or Unloading of Liquid Bulk	Regulated	Per ton			5.65	1.02	2			
Section 5.2	Special Services - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
5.2.1	Use of Contention Barriers										
5211	Use of Contention Barriers (Tariff per day or fraction day) (n73)	Regulated	Per day			1 758 00	316 44	1		1 590 00	286.20

(n73). This is applicable for the use of contention barriers during loading and unloading operations of hazardous substances and/or goods such as hydrocators and other liquid bulk. Services includes plexing and unplicing contention barriers as well as cleaning and dying of contention barriers. Tailf of this service during loading juniceding operations of hazardous substances and/or goods such as hydrocators and other liquid bulk. Services includes plexing and unplicing contention barriers as well as cleaning and dying of contention barriers. Tailf of this service during loading juniceding operations of hazardous substances and/or goods such as hydrocators and other liquid bulk. Services includes plexing and unplicing and

SECTION 6		PASSENGERS									
Section 6.1	Standard Service	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
6.1.1	Standard Service to the Vessel - Services to the Vessel										
6.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.69	0.30						

19.98

3.60

 6.1.1.1 Use or rental of berth (Tariff per hour or fraction hour) (n1)
 6.1.2 Standard Service per Passenger
 6.1.2.1 Standard Service per Passenger (n74)
 (n74): It applies to every passenger embarked or disembarked; if the same passenge Per Passenger Regulated ers are not subject to charge encer disembarks and disembarks (on the same or different vessel) the tariff is acciled each time. Children under 12 years old and crew m

SECTION 7	OTHER COMMON SERVICES TO THE VESSEL (n75)										
Section 7.1	Special Services - Services to the Vessel	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
7.1.1	Use of Contention Barriers										
7.1.1.1	Use of Contention Barriers (Tariff per hour or fraction hour) (n76)	Regulated	Por día	1,758.00	316.44			1,590.00	286.20		
7.1.2	Hold Cleaning (n77)										
7.1.2.1	Vessel Holds cleaning (price per gang-shift or fraction) (n78)		Per gang/Shift	510.00	91.80						
7.1.2.2	Unloading of non-hazardous solid waste (price per M3 or fraction) (n79)	Non Regulated	Per M3	255.00	45.90						
7.1.2.3	Unloading of hazardous solid waste (price per M3 or fraction) (n80)		Per M3	400.00	72.00						
7.1.3	Sludge Management										
	Oily Waste Management (n81)		Per truck	1,326.00	238.68						
7.1.3.2	Unloading of solid waste (price per M3 or fraction) (n82)	Non Regulated	Per M3	357.00	64.26						
7.1.3.3	Management of non-Hazardous solid waste - non recoverables (price per M3 o fracción) (n83)	Hon Hogalatoa	Per M4	357.00	64.26						
7.1.3.4	Management of Hazardous solid waste (price per M3 or fraction) (n84)		Per M3	500.00	90.00						
7.1.4	Other services										
7.1.4.1	Water supply (n85)	Non Regulated	Per ton	10.60	1.91						
7.1.4.2	Phone service	Non Regulated	Per activity	Per q	linote						

(n75) Services applicable to all type of vessels. (n76): This is applicable for the use of contention barriers during fuel supply operations to the vessel. Service includes placing and unplacing contention barriers as well as cleaning and drying service. Tariff for contention barriers shall be applicable to the shipping line or its representative (n77): This service will apply to users named "bients of ordinary service" which refers to loading or discharging cargo not compatible and ading operations of compatible goods per vessel call. Service shall not be applicable to discharging and loading operations of incompatible goods of same vessel call, in other words, to discharging cargo not compatible with products to be loaded. This includes vessel hold determing service process, and it does not include washing of holds.

(n78): This service refers to picking up solid wastage onboard the vessel such as materials from lashing/unlashing (wood, cardboard and others) as well as other authorized solid wastage (wastage of shipped or unloaded cargo, etc.).

(17)? This refers to unloading solid westage onboard the vessel such as materials from Eshinguinitabing (wood, cardboard and others) as well as front westage (westage of shipped or unloaded cargo, etc.) not classified as dangerous. Service includes temporary store in the terminal facility if necessary, as well as final weste disposal. Minimum invoice shall be USD 255. (n80): This refers to unloading hazardous solid westage onboard the vessel from the dearing of the ship's holds and final disposal of westes from vessel. Minimum invoice shall be USD 400 (n81): Price of this service includes collection and final disposal of oily waste from vessel. (#2): This service refers to discharging or collecting from vessels recoverable waste from vessels (such as: plastics, plastics, metal costs, wood, cardboard, paper, magazines and any other of the same type), classified as non-dangerous. Price of service applies to those cases where APM Terminals Callao S.A. does not carry out vessel hold clearing service. Minimum invoice shall be USD 357. (n3): This service refers to discharging or collecting from vessels non-recoverable waste classified as non-dangeorus. Service indudes temporary store in the terminal facility if necessary, as well as final waste disposal. Price of service applies to those cases where APM Terminals Callao S.A. does not carry out vessel hold dearing service (section 7.1.2). Minimum invoice shall be USD 357

(nP4): This service refers to discharging or collecting from vessels hazardous solid wastage. Service includes temporary store in the terminal facility if necessary, as well as final waste disposal. Price of service applies to those cases where APM Terminals Callao S.A. does not carry out vessel hold dearing service (section 7.1.2). Minimum invoice shall be USD 500.

(n85): This shall be provided according to the Terminal's berth/pier availability or capacity and shall be charged to the shipping line or representative

SECTION 8	COMMERCIAL POLICIES (#6)										
Section 8.1	Cancellation of loading	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
8.1.1	Cancellation of container loading and withdraw of container (n87)										
8.1.1.1	20' Container	Non Regulated	Per container			217.00	39.06				
8.1.1.2	40' Container	Non Regulated	Per container			272.00	48.96				
812	Cancellation of break bulk and ro-ro cargo loading (p88)										

(65): For further details on the scope and conditions of applicable discounts, of schera and promotions, see Charter V of Tariffs Regulation and Commercial Policy of APM Terminals Callao SA published on website: www.apmleminalscallao.com.pe

(167): This is applicable whenever the user (consignee, representative or shipping line) withdraws a container after cancellation and container withdraw. In this case standard service tariff shall not be applicable. It applies for full and empty containers. Tariffs of section 1.2 (for empty) or 1.3 (for full) that refer to the use of the operational area applied if the container stays more than 48 hours.

stay longer than the free time granted by the standard service to these kinds of cargo

Section 8.2	Berthing window reservation	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	
8.2.1	Contracting of berth window reservation for vessels											
8.2.1.1	Contracting of guaranteed berth window reservation for regular traffic vessels with ro-ro cargo	Non Regulated	by vessel landfall	10,000.00	1,800.00							

SECTION 9	SURCHARGES (#88)							
Section 9.1	General surcharges	Unit of charge	USD (\$)	IGV	Total			
9.1.1	Certificates and receipts							
9.1.1.1	New receipt printing	Per receipt	10.00	1.80	11.80			
9.1.1.2	Issue of weight certificate	Per document	5.00	0.90	5.90			
9.1.1.3	Issue of copy of weighing ticket and/or tally note	Per document	5.00	0.90	5.90			
9.1.1.4	Letter of amendment and/or sending new details to SUNAT	Per document or new transmission	20.00	3.60	23.60			
9.1.1.5	New collection settlement (n90)	Per document	8.00	1.44	9.44			
9.1.2	Returned checks and default (n91)							
(n89): For further i	information and scope, refer to Charter V, section 5.6 of the Tariffs Regulation and Commercial Policy of APM Terminals Callao SA. published on website: www.apmterminalscallao.com.pe							
(n90): This applies	when a collection settlement is generated more than once by the user.							
(n91): Surcharge n	to subject to IGV. See section 5.6 f the Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A.							
Section 9.2	Surcharges applicable to the Vessel	Unit of charge	USD (\$)	IGV	Total			
9.2.1	Surcharges for non-use or non-release							
9.2.1.1	Offset for not used crew (n92)	Per crew/hour	162.00	29.16	191.16			

9.2.1.2 No release of berth Per hour 600.00 Not subject to IGV 600.00 9.2.2 Liquid Spill Management Surcharges 
 9/2.2
 Liquid optimizangement, ownerwayee
 Per case
 Subject to price quote

 9/2.1
 Liquids plim management, ref(s)
 Per case
 Subject to price quote

 (r82): The unit of charge applies to every not used creat/hour or creatilization of an hour.
 Per case
 Subject to price quote

 (r83): This applies when liquids are splited on the water area or quay of the Multipurpose North Terminal due to user's responsibility, it indudes removing and/or collecting and final disposal of the substances, in accordance with current regulations. This surcharge will apply only when APMT carries out the split

Section 9.3	Surcharges applicable to the Cargo	Unit of charge	USD (\$)		Total	
9.3.1	Change of status					
9.3.1.1	Change of container status	Per container	50.00	9.00	59.00	
9.3.1.2	Change of non-containerized cargo status	Per B/L	108.60	19.55	128.15	
9.3.2	Late arrival					
9.3.2.1	Late arrival of containers	Per container	200.00	36.00	236.00	
9.3.2.2	Late arrival of break bulk cargo	Per ton	2.40	0.43	2.83	
9.3.2.3	Late arrival of Ro-ro cargo	Per unit	150.00	27.00	177.00	
9.3.2.4	Late arrival of solid bulk cargo	Per unit	1.50	0.27	1.77	
9.3.2.5	Late arrival of liquid bulk cargo (n94)	Per unit	1.00	0.18	1.18	
9.3.3	Late submission					
9.3.3.1	Late submission of shipping documents of containers (n95)	Per DAM	85.00	15.30	100.30	
9.3.4	Missing, cancelling or rescheduling appointments (n96)					
9.3.4.1	Missing an appointment (n97)	Per appointment	16.00	2.88	18.88	
2.3.4.2	Cancelling or rescheduling an appointment (n98)	Per appointment	10.50	1.89	12.39	
9.3.5	Rescheduling services					
9.3.5.1	Rescheduling movement for containers (n99)	Per container	100.00	18.00	118.00	
9.3.6	Other surcharges to the Cargo					
9.3.6.1	Bulk cargo re-delivery to truck (n100)	Per truck	225.00	40.50	265.50	
9.3.6.2	Liquid spill management (n93)	Per case Por cotización				
	s only to shipping through tanker trucks.					
(n95): This applie	s only to the document submission of containers that are taken into the Temporary Depot (3014) out of the cut off set by APM TERMINALS CALLAO S.A.					

(n96): Appointments will last one (01) hour

(n97): Missing an appointment means a user does not attend within the period granted by APM TERMINALS CALLAO S.A., and also when the user tries to cancel or reschedule its appointment during the period between one hour and one minute before the appointment

(r08): The cancellation surcharge applies when the user cancels the appointment within eight and one hours before the start of the appointment. The rescheduling surcharge applies when the user reschedules the appointment within eight and one hours before the start of the first appointment. No cancellation or rescheduling surcharge will be charged if the user does it eight (08) hours before the appointment.

(r/9): The surcharge applies when the customer or its representative, after failing to attend or cancelling a scheduled and approved container movement service (related to an inspection, consolidation/seconsolidation or other service), requests the rescheduling of such service. The user requests revice (related to an inspection, consolidation/seconsolidation or other service), requests the rescheduling of such service. The user requests revice (related to an inspection, consolidation/seconsolidation or other service), requests the rescheduling of such service. The user requests revice (related to an inspection, consolidation/seconsolidation or other service), requests the rescheduling of such service. The user requests are for the formation in the service (related to an inspection). Note that, according to the Operations Regulations, the deadline to cancel a mobilization service is between the request such service. The user requests two reschedulings for the same container the user will be charged builts the sucharge). Examples of services that require container mobilization: customs inspection (by red channel), SENASA BOE (baccordin), inventory, consolidation/deconsolidation, among others.

(n100): This applies when solid cargo fails to the ground while uploading it from silos to truck or from hopper to truck due to driver's or truck's inefficiency (e.g. not closing truck's door property). It includes siab cleaning and bulk cargo re-delivery to truck. Surcharge will be charged to the user (consignee, cus

COMMERCIAL POLICIES

## DISCOUNTS, OFFERS AND PROMOTIONS

Discounts, offers and promotions are commonly accepted practices that answer to compensatory circumstances which shall be determined by APM TERMINALS CALLAO S.A. in compliance with the principle of non-discrimination, equally, neutrality, prohibition of price transference and free election. In this regard, APM TERMINALS CALLAO S.A. shall generally establish the following commercial practices of discounts, offers and promotions:

Promotions and/or temporary offers
PAPITERMINALS CALLAO S.A. may temporarly provide, as part of its commercial policy, offers and/or promotions, as long as permanent equivalent conditions remain.
For cases foreign in Section 5.1 and 5.1.2 of Taim's Regulation and Commercial policy, offers and/or promotions, and/or offers with details for the adnowledgement of users.
PAPITERMINALS CALLAO S.A. tail and subtain humation due to modification of discours, pormotion and/or offers with details for the adnowledgement of users.
PAPITERMINALS CALLAO S.A. tail and subtain humation due to modification of discours, pormotions, pormotion and/or offers and/or promotions.
CONTENT APPL CATION CF DISCOURTS, OFFERS AUD PROMOTIONS
PAPITERMINALS CALLAO S.A. tail and provide adsounds, offers and/or promotions to fees of current Tainfa and/or Prices applicable to the Vessel and/or the Cargo considering the following oiteria:
Combrande available
Partners March Case and/or additional devices growted without the application of discourts and/or promotions.
Generation of more themated devices growted to Vessels, Cargo and others.
Generation for more discourds, discourds, offersion, default and/or promotions will be established according to cocking the query cargo volume and other ories that are applicable by its nature.
In the event of Services provided to the Cargo, discounds, offersion and/or growtex will be established according to colors growtex and other ories in that are applicable by its nature.
Discounds, offersia and/or promotions will be established according to colors provide to advice grow volume and other ories in that are applicable by its nature.
In the event of Services provided to the Cargo, discounds, offersia and/or promotions will be established according to colors provide to advices provided by its nature.
Discounds, offersia and/or promotions will be established according to colors provide to advice grow volume and other ories in that are applicable by its nature.

### SPECIAL SERVICES WITH DISCOUNTS FIXED IN THE CONCESSION AGREEMENT

ion Contract with a hundred per cent discount (100%), whenever the User requests this as complementary service to the Standard Servic

SPECIAL SERVICES WITH UISCOUNT 5 YRED IN THE CONCESSION AGREEMENT APM TERMINUES CLUID 63. And in provide the following Special Services established in the Concess Hardn movilation (ISO Hatches) Restoraged containers: Flod or via quary) Shippinglunkading and delivery to (SOUCOG (Additional) - container 20 and 40 foot Horizontal movement in the Maliproprox horizontal Terminal Handling of Compo per registry (with or without forkiths) Restore entropy Reefer energy Reefer inspection and monitoring Assembling and disassembling of clip system on reefer Gensets

ipecial Services shall be provided considering principles set forth in the Tarif's Regulation and Commercial Policy, and the Concession Contract, so APM TERMINALS CALLAO S.A. and the User may not bridge rules and practices that may alter competition of Terminal Services, or that implies abusive and ineffic us. For such reason, above detailed port services shall be provided as per regular and reasonable operations that Foreign Trade demands.

Description details of each service are included in Chapter VII of Services Summary of the Tariff's Regulation and Commercial Policy.

# RESERVE OF GUARANTEED BERTHING WINDOW

Space may be reserved for Vessels dooling provided that this has been previously anranged between APM TERMINALS CALLAO S.A. and the shipping lines as per terms and conditions set forth in the Operations Regulations and procedures from the Berthing Window Reservation for Regular Service Vessels, established within Causes 3.14 of the Concession Contract.

# APPLICATION OF TAX ON GENERAL SALES (IGV)

The rates and prices of standard and special services taxed at zero percent (0%) rate, which by version 4.0 of Tariff Scheme were listed as exempt from VAT, refer to those services according to the VAT Law considered as exports of services. Rate of 0% will apply as long as the User is con Cargo Carrier<sup>®</sup>, Other than that the current rate of VAT amounting to 15% will apply.

PATMENT CONDITIONS Currency of Payment: Tariff and prices in American Dolars (US\$) may be canceled in Nuevos Soles (S<sup>2</sup>) according to the average sale exchange rate of the financial system indicated by Superintendence of Bank, Insurance and AFP (SBS) at the date of payme

Form of Payment: Payments may be through bank account transference, deposit or by check in the same bank that for this purpose will be assigned by APM TERMINALS CALLAO S.A.

Date of Payment: Payments shall be done according to the following considerations:

### OFFERS AND REDUCTIONS OF TARIFFS AND PRICES

Offers or tain and/or prices reductions apply to individual companies, but also be commonic groups, in order for the offer or discount to apply to the companies forming an economic group, the following conditions must be net, at least. (i) that the principal or parent company is the one requesting the offer, and (ii) that the principal or parent company is the one requesting the offer, and (ii) that the principal or parent company is the one requesting the offer, and (ii) that the principal or parent company, and the second is a common group. The learns of which company or companies will be inviced will be coordinated with the main or parent company.

e draged to strange ase completed days off use of the operating area that and pornotion grants are not retracked. For example, lem high off the 11 grants table of 20 fee days, if the user remains in the atorage area until days 35, the prior of subaction 1.3.1.7 of the TailTL table has get mode 50 and 50. A prior at absorbing 1.2 of the table table of the ange to mady 1.2 to be strate table of 20 fee days, if the user remains in the atorage area until days 35, the prior of subaction 1.3.1.7 of the TailTL table the ange to mady 1.2 days 3. total of 10 free days: if the user's carry

# (Group 1) SPECIAL OFFER FOR DRY CARGO CONTAINERS IN APM TERMINALS CALLAO'S TEMPORARY DEPOT

#### Dear customers and users

APVT Termines Calacy CPMIC() has as topocal offers for the containers processed as Temporary Depo (code 3014) that meet the requirements below, according to the type of operation. Offers indude the extension of days free of use of operational area and/or discusts on the price of the integrated temporary deposit service. Each promotion is independent (not cumulative); for example, if the user discuss to the type of operational area and/or discuss to meet the price of the integrated temporary deposit service. Each promotion is independent (not cumulative); for example, if the user discuss to meet the type of operational area and/or discuss to meet the price of the integrated temporary deposit service. Each promotion is independent (not cumulative); for example, if the user discuss to meet the type of the integrated temporary deposit service. Each promotion is independent (not cumulative); for example, if the user discuss to meet the type of the integrated temporary deposit service. Each promotion is independent (not cumulative); for example, if the user discuss to meet the type of the integrated temporary deposit service. Each promotion is independent to the type of the integrated temporary deposit service. Each promotion is independent to the type of the integrated temporary deposit service. Each promotion is independent to the type of the integrated temporary deposit service. Each promotion is independent to the type of the integrated temporary deposit service. Each promotion is independent to the type of temporary deposit service. Each promotion is independent to the type of temporary deposit service. Each promotion is independent to the type of temporary deposit service. Each promotion is independent to the type of temporary deposit service. Each promotion is independent to the type of temporary deposit service. Each promotion is independent to the type of temporary deposit service. Each promotion is independent to the type of temporary deposit service. Each promotion is independent to the type of temporary d notion 1.1 they would

#### 1.1 Discount on Integrated Service Price and extension of free days - dry cargo import

- Target group: Importers of dry cargo containers (consignees or its represe tatives -customs brokers, logistics operators, etc.).

- Segment: Importers who address dry cargo to APMTC's Temporary Depot.

-Temporary officr: Importers of dry cargo (consignees or its representatives --oustoms brokers, logistics operators, etc.) who guarantee minimum amounts equal to or greater than 200, 300, 800, 1.5 thousands and 3 thousand containers through the APHITC's Temporary Depot will be granted discounts and extensions of the operating area (laking into account the 10 days including in the Integrated Service of Import Temporary Depot, sections 1.4.1.1 and 1.4.1.2 of the Tariff Lady as below:

(i) ≥ 200 to 299 containers = 10 extra free days (resulting in a total of 20 free days)

(i) > 300 to 799 containers = USD 40 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers and 10 extra free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tariff (reaching a total of 20 free days)

(iii)  $\geq$  800 to 1,599 containers = USD 40 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers and 20 extra free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tariff (reaching a total of 30 free days)

- (iv) > 1.600 to 2.999 containers = USD 50 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers and 20 extra free days to those already oranted in subsections 1.4.1.1 and 1.4.1.2 of the Tariff (reaching a total of 30 free days)
- (v) ≥ 3.000 containers = USD 60 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers and 20 extra free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tariff (reaching a total of 30 free days)

# - Effective: Importers may apply or request this promotion until June 30, 2024.

The minimum volumes of the offer will be accounted for annually. That is, in the 12 months following the date of the agreement (between APMIC and the user) to start accounting for the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed to start accounting on February 1, 2024, then the accounting will be made until January 31, 2025). For this purpose, users must submit to the commercial area of APMIC the projections of containers to be imported in the annual period indicated above, in order to determine the availability of space at the terminal.

APMTC will carry a review every three (03) months of the imported volume. It is observed that in the first quarter at least 45% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled. This consideration applies to offers whose movement accounting starts on or after January 1, 2024.

For offers already agreed before January 1, 2024, the genalties described in version 12.5 or earlier of the Tariff List apply.

Acceptance to any of the offers rances (whether i. ii. iii. iv or v) will be subject to an assessment of the orojections and/or that the user has mobilized by the Port of Calac. in the vear preceding the agreement to start accounting, at least 80% of the volume committed

In the event that the offer has been cancelled to a user and the user requests access to a new offer (whether 1, ii, ii, or v), the new access will be granted if the following condition is met: that during the period of the cancelled offer (due to failure to comply with the volume commitment between the accounting start date and the cancellation date) it is verified that this user has handled dire by DT 3014 equal to or more than 80% of the total containers that have been directed through a temporary deposit (3014 or extracord) and unceded/caded by North Multi-purpose Terminal.

In case of changes in the prices of the subsections 1.4.1.1 of 1.4.1.2 of the Tariff List, the discounts of USD 40, USD 50 or USD 60 apply to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tariffs and Commercial Policy of APM Terminals Calao.

# - This offer is subject to availability of space determined by APMTC.

## 1.2 Discount on Integrated Service Price and extension of free days - dry cargo export

- Target group: General users (exporters, forwarders or its representatives -customs brokers, logistics operators, etc- and other users) of containers with dry export cargo.

# - Segment: Exporters who address dry cargo to APMTC's Temporary Depot.

- Temporary offer: Exporters (forwarders or its representatives -outdoms brokers, looispics operators, etc.) who quarantee minimum amounts of more than 300.700, 800, and 3000 containers using the APMTC's Temporary Decot will be granted discounts and extensions of the operating area use (taking into account) the 7 days including in the Integrated Service of Export Temporary Decot will be started discounts and extensions of the operating area use (taking into account) the 7 days including in the Integrated Service of Export Temporary Decot sections 1.4.2.1 and 1.4.2.2 of the Tariff List), according to the table below:

(i) ≥ 300 to 699 containers = 3 extra free days (resulting in a total of 10 free days)

(ii) ≥ 700 hasta 799 contenedores = USD 20 discount from the price set out in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers and 3 extra free days (resulting in a total of 10 free days)

(ii) ≥ 800 to 2,999 containers = USD 20 discount from the price set out in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers and 8 extra free days (resulting in a total of 15 free days)

(v) ≥ 3,000 containers = USD 20 discount from the price set out in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers and 13 extra free days (resulting in a total of 20 free days)

# - Effective: Exporters may apply or request this promotion until June 30, 2024.

The minimum volumes of the offer will be accounted for annually. That is, in the 12 months following the date of the agreement (between APMIC and the user) to start accounting for the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed to start accounting on February 1, 2024, then the accounting will be made until January 31, 2025). For this purpose, users must submit to the commercial area of APMIC the projections of containers to be imported in the annual period indicated above, in order to determine the availability of space at the terminal.

APRIC dut cany a review every these (0)) months of the approted volume. If is a beamed that is he far quarter at least 20% of the committed annual volume has not been met, the Offer all be canceled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met, the Offer all be canceled; and if the review of the third quarter finds that at least 20% of the committed annual volume has not been met. the Offer all be canceled; and if the review of the third quarter finds that at least 20% of the committed annual volume has not been met. The Offer all be canceled; and if the review of the third quarter finds that at least 20% of the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer all be canceled; the committed annual volume has not been met. The Offer

For offers already agreed before January 1, 2024, the penalties described in version 12.5 or earlier of the Tariff List apply.

Acceptance to any of the offers ranges (whether i, ii, iii or iv) will be subject to an assessment of the projections and/or that the user has mobilized by the Port of Calao, in the year preceding the agreement to start accounting, at least 80% of the volume committed.

In the event that the offer has been anothed to a user and be user requests accounting start date and the cancellation date) it is wrifted that this user has hard TO 3104 equal to or more hand 5% of the cancellation date is a the part of the cancellation date) it is wrifted that this user has hard TO 3104 equal to or more hand 5% of the concellation date is a the part of the cancellation date) it is wrifted that this user has hard TO 3104 equal to or more hand 5% of the concellation date) it is wrifted that this user has hard to 2104 equal to or more hand 5% of the been directed through a date and the cancellation date) it is wrifted that this user has hard TO 3104 equal to or more hand 5% of the been directed through a date more directed through a date and the cancellation date) it is wrifted that this user has hard TO 3104 equal to or more hand 5% of the been directed through a date and the cancellation date) it is wrifted that this user has hard TO 3104 equal to or more hand 5% of the been directed through a date and the cancellation date) it is wrifted that this user has hard TO 3104 equal to or more hand 5% of the been directed through a date and the cancellation date) it is wrifted that this user has hard the cancellation date and the cancellation date and the cancellation date) it is wrifted that this user has hard through a date and the cancellation date and the can

This offer is subject to availability of space determined by APMTC

### 1.3 Discount on Integrated Service Price and extension of free days - Import/Export of dry cargo

Target group: General users (importensionsignees and exportensionwarders or its representatives -customs torkies, logistics operators, etc.- and other users) of containers with importeuport dry cargo. Timport containers only are mobilized, the discount on item 1.1 will be granted, and if export containers only are mobilized.

### - Segment: Users who address dry cargo to APMTC's Temporary Depot.

Temporary offer: Users who mobilize import and export dry cargo containers together who guarantee minimum amounts equal or more than 300, 700, 800, 1600 and 3,000 con ervice for Import and Export Dry Containers (Sections 1.4.1.1, 1.4.1.2, 1.4.2.1 and 1.4.2.2 of the Tariff List), according to the table below: ners through the APMTC's Temporary Depot will be granted disc nal area with respect to the prices and days of use of free op ating area considering in the Special Inte nal free days with respect to the free days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export co

(i) 2 300 to 699 containers = 10 additional free days with respect to the free days of use of operational area indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (which results in a total of 20 free days) and 3 add

The capity of a containers = USD 40 discount and 10 additional free days with respect to the prices and free days indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for report containers (resulting in a total of 20 free days) and USD 20 discount and 3 additional free days with respect to the prices and days indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for report containers (resulting in a total of 10 free days) and USD 20 discount and 3 additional free days with respect to the prices and days indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 10 free days) and USD 20 discount and 3 additional free days with respect to the prices and days indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 10 free days) and USD 20 discount and 2 additional free days with respect to the prices and days indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 10 free days) and USD 20 discount and 2 additional free days with respect to the prices and days indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 10 free days) and USD 20 discount and 2 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for report containers (resulting in a total of 15 free days). (iv) > 1,600 to 2,999 containers = USD 50 discount and 20 additional free days with respect to the prices and fays indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 discount and 8 additional free days with respect to the prices and days indicated in sections 1.4.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 discount and 8 additional free days with respect to the prices and days indicated in sections 1.4.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 discount and 8 additional free days with respect to the prices and days indicated in sections 1.4.1 and 1.4.1.2 of the Tariff List for export containers (resulting in a total of 130 free days) and USD 20 discount and 8 additional free days with respect to the prices and days indicated in sections 1.4.1 and 1.4.1.2 of the Tariff List for export containers (resulting in a total of 15 free days)

(v) > 3,000 containers = USD 60 discount and 20 additional free days with respect to the prices and days indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 discount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 20 free days) and USD 20 discount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 20 free days) and USD 20 discount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 20 free days) and USD 20 discount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 20 free days) and USD 20 discount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 20 free days) and USD 20 discount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 20 free days) and USD 20 discount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 20 free days) and USD 20 discount and 13 additional free days with respect to the prices and tariff List for export containers (resulting in a total of 20 free days) and USD 20 discount and 13 additional free days with respect to the prices and tariff List for export containers (resulting in a total of 20 free days) and USD 20 discount and 13 additional free days with respect to the price

#### Effective: Users may apply or request this offer until June 30, 2024. - Requirements:

The minimum volumes of the offers will be accounted for annually, that is to say, in the 12 months tolowing the date of the agreement (between APMITC and the user) to start counting the volumes (for example and as a hypothetical case, if the user's request for promotion is accepted and it is agreed that the posting on February 1, 2024, then the accounting will be made until January 31, 2025). For this purpose, users must submit to the commercial area of APMITC the projectors of containers that will be imported and exported in the annual period indicated above, in order to determine the availability of space in the terminal.

APINTC will carry a review every three (IC) months of the imported/exported volume. It is observed that in the first quarter at least 20% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 20% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 20% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 20% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 20% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 20% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 20% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 20% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 20% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the third quarter finds that at least 20% of the review of the review of the third quarter finds th

For offers already agreed before January 1, 2024, the genalties described in version 12.5 or earlier of the Tariff List apply.

In the set that the drive has been cancelled to a user and he user regresses to a new differ (has in the following condition is and that drive drive has been drived through a set of the cancelled drive (has to failure to comply with the volume commitment between the accounting start data and the cancellation data) it is writted that this user has handleddirected by 107 314 (equal to row than 30 and the cancellation data) and the cancellation data) it is writted that this user has handleddirected by 107 314 (equal to row than 30 and the data containes that that the data containes that the been fielded through a law provide provide data and the cancellation data) it is writted that this user has handleddirected by 107 314 (equal to the cancellation data) and through a period of the cancellation data).

In case of changes in the prices of the subsections 1.4.1 or 1.4.2 of the Tariff List, the discounts apply to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tariffs and Commercial Policy of APM Terminals Calao

- This offer is subject to availability of space determined by APMTC.

#### 1.4 Discount on the price of integrated service and extension of free days - Import/Export of dry cargo (only for new customers and without volume commitment)

- Target group: General users (importensionsignees and exportensionwarders or its representatives -customs brokens, logistics operators, etc. and other users) of containers with importiespot dry cargo. This offer is independent of the offers listed above (1.1 to 1.3); that is, those who are granted this promotion will not be eligible for offers from 1.1 to 1.3. Applies to new users

In case of changes in the prices of the subsections 1.4.2.1 or 1.4.2.2 of the Tariff List, the discounts of USD 20 apply to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tariffs and Comme

- Temporary offer: Users who mobilize import and export dry cargo containers who guarantee minimum amounts equal or more than 150, 400, 800 and 1,500 containers through the APMITC's Temporary Depot will be granted a discount on the price of the Special Integrated Temporary Depot Service for Import and Export Dry Containers (Sections 1.4.1.1, 1.4.1.2, 1.4.2.1 and 1.4.2.2 of the Tariff List) and an ex-of days free of use of operational area, according to the table below.

(i) ≥ 150 to 39 or containers = For import containers a discourt of USD 40 on the price indicated in sections 14.1.1 and 14.1.2 of the Tariff List and 10 free days in addition to those elseady granted in section Tariff List and The days in addition to those elseady granted in sections 14.2 1 and 14.2.2 the Tariff List (reaching a blaid of 10 free days).
(ii) ≥ 400 to 799 containers = For import containers a discourt of USD 40 on the price indicated in sections 14.1.1 and 14.1.2 of the Tariff List and 10 free days).
(iii) ≥ 400 to 799 containers = For import containers a discourt of USD 40 on the price indicated in sections 14.1.1 and 14.1.2 of the Tariff List and 20 free days in addition to those elseady granted in sections 14.2.1 and 14.2.2 the Tariff List and Tariff List and 750 explicit and 20 free days in addition to those elseady granted endscale to 14.2.1 and 14.2.2 the Tariff List and 750 explicit and 20 free days). ted in sections 1.4.1.1 and 1.4.1.2 of the Tariff List (reaching a total of 20 free da unt of USD 20 on the price indicated in se ons 1.4.1.1 and 1.4.1.2 of the Tariff List (reaching a total of 30 free days); while for exp

ns 1.4.1.1 and 1.4.1.2 of the Tariff List (reaching a total of 30 free da

(iii) ≥ 800 to 1,499 containers = For import containers a discount of USD 50 on the price indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List and 20 free days in addition to those already granted in sections 1.4.2.1 and 1.4.2.2 of the Tariff List and 8 free days in addition to those already granted in sections 1.4.2.1 and 1.4.2.2 of the Tariff List and 8 free days in addition to those already granted in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 15 free days). (n) ≥ 1,500 containers = For import containers a discount of USD 60 on the price indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List and 20 free days in addition to those already granted in sections 1.4.1.1 and 1.4.1.2 of the Tariff List (reaching a total of 30 free day List and 13 free days in addition to those already granted in sections 1.4.1.1 and 1.4.2.2 of the Tariff List (reaching a total of 20 free days). nt of USD 20 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff

# - Effective: The offer is valid from January 3, 2024 to June 30, 2024.

Requirements: The volumes of the promotions will be accounted for during the period of validity of the offer, that is, between January 3, 2024 and July 2, 2024. For these offers there is no prior agreement or commitment of movements by the user vis-à-vis APM Terminals Calao.

#### (Group 2) PROMOTION FOR REFRIGERATED CARGO CONTAINERS IN APM TERMINALS CALLAO

APM Terminals Calao (APMTC) has arranged offers for containers ful with refrigerated cargo to be served by APM Terminals Calao (APMTC), either as a Port Terminal or through our Temporary Depot (code 3014), and that meet the considerations indicated ines below, depending on the type of operation

#### 2.1 For Export containers using APMTC as Temporary Depot

- Target group: Exporters (forwarders or its rep esentatives -customs brokers, logistics operators, etc. and other users) of containers with refrigerated cargo (reefer container)

and any other. Refrigerated cargo exporters who insure a minimum annual volume of 1,000 refer containers through APMIC's Temporary Depot will be exempt of the late-delivery container shipping document surcharge (do

# - Effective: Users may apply or request this offer from May 02, 2024 until June 30, 2024.

- Requirements

The minimum volumes of the offer will be accounted for annually, that is to say, in the 12 months following the date of the agreement (between APMITC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the posing on June 1, 2024, then the accounting will be made until May 31, 2025). For this purpose, users must submit to the commercial area of APMITC the projections of reefer containers that will be exported in the annual period indicated above, in order to determine the availability of space in the terminal.

APRIC and garry a meter early there (02) months of the volume of containers motivated by the same where the first is determined during the meter that the user will not be able to meet the required and user. the APRIC may cancel the ponotion. If all lead 25% of the minimum annual volume required is not reached in the first leview, the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled in the first leview, the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled. If a be able to meet the temporary offer will be canceled. If a beat to meet the temporary offer will be canceled. If a beat to meet the temporary offer will be canceled. If a beat to the meet temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled to temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled. If a beat to temporary offer will be canceled to temporary offer will be canceled. If a beat to temporary offer will be canceled to temporary offer will be cancele elled; if at

### - This offer is subject to availability of space determined by APMTC.

#### 2.2 For import containers that use APMTC as temporary depot (with a maximum of 8 days of stav)

Target group: Importers of dry cargo containers (consignees or its representatives -customs brokers, logistics operators, etc.) of containers with refrigerated cargo (reefer containers) that who address dry cargo to APMIC's Temporary Depot

- Tempcorp dfr: "Importes who guarantee a minimum annum movement of 200 containes (40 feet) or refigurated cargo will be granted a discount of USD 38.5 on the current pices, at the time of invoicing, for the package of services consisting of the integrated tempcorp storage service for import refer containers (tablection 1.4.3.2), day 8 of peetional area use (subsection 1.4.3.3), day 8 of refer power supply (under the first or the package of services consisting of the integrated tempcorp storage service for import refer containers (tablection 1.4.3.2), day 8 of refer power supply (under the first or the package of services consisting of the integrated tempcorp storage service for import refer containers (tablection 1.4.3.2), day 8 of refer power supply (under the first or the day 6 of importion of monohysis (tablection 1.5.4.2). - Effective: The offer may be requested from August 02, 2023 to June 30, 2024.

- Requirements:

The minimum volumes will be counted annually, that is to say, in the 12 months following the date of the agreement (between APMITC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the posting on September 1, 2023, then the accounting will be made until August 30, 2024). For this purpose, users mu to the commercial area of APMITC the projections of containers with import refrigerated case to that when a 173014 in the annual period indicated advine, in order to determine the availability of space in the terminal.

APRIC out gray a seve every three (13) months of the insported volume. If it is observed that in the fart quarter at least 20% of the committed annual volume has not been met, the Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met, the Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met. The Offer will be cancelled; if it is observed that the second quarter at least 45% of the committed a For offers already agreed before January 1, 2024, the penalties described in version 12.5 or earlier of the Tariff List apply

The acceptance of an request for access to the offer will be subject to an evaluation of the projections and/or that the user has mobilized by the Port of Calao, in the year preceding the agreement to start accounting, at least 80% of the volume committed.

In the event that the offer has been cancelled to a user and the user requests access to a new offer, the new edges will be granted if the following condition is met: that during the period of the cancelled offer (due to failure to comply with the volume commitment between the accounting start date and the cancellation date) it is verified that this user has handled/seded by IOT 3014 equal to or more than 80% of the total containers that have been directed through a temporary deposit (2014 or extraport) and unbaded/seded by North Multi-purpose Terminal.

- This offer is subject to availability of space determined by APMTC.

# ers that use APMTC as temporary depot (with a maximum of 11 days of stay)

- Target group: Importers of dry cargo containers (consignees or its representatives -customs brok ers, logistics operators, etc.) of containers with refrigerated cargo (reefer containers) that who address dry cargo to APMTC's Temporary Depot

- Temporary offer: Importers who guarantee a minimum annual movement of 150 containers (40 feet) of refrigerated cargo will be granted a discount of USD 653.6 on the current prices, at the line of invoiding, for the package of services consisting of the integrated temporary storage service for import neeler containers (40 feet) of refrigerated cargo will be granted a discount of USD 653.6 on the current prices, at the line of invoiding, for the package of services consisting of the integrated temporary storage service for import neeler containers (40 feet) of refrigerated cargo will be granted a discount of USD 653.6 on the current prices, at the line of invoiding, for the package of services consisting of the integrated temporary storage service for import neeler containers (40 feet) of refrigerated cargo will be granted a discount of USD 653.6 on the current prices, at the line of invoiding, for the package of services consisting of the integrated temporary storage service for import neeler containers (40 feet) of refrigerated cargo will be granted a discount of USD 653.6 on the current prices, at the line of invoiding, for the package of services consisting of the integrated temporary storage service for import neeler containers (40 feet) of refrigerated cargo will be granted a discount of USD 653.6 on the current prices, at the line of invoiding, for the package of services consisting of the integrated temporary storage service for import neeler containers (40 feet) of refrigerated cargo will be granted a discount of USD 653.6 on the current prices, at the line of invoiding, for the package of services consisting of the integrated temporary storage service for import neeler containers (40 feet) of the line of integrated temporary storage service for import neeler containers (40 feet) of the line supply (sub ection 1.5.2.3) and days 8 to 11 of inspection and monitoring (subsection 1.5.2.4) Effective: The offer may be requested from May 02, 2024 to June 30, 2024

The minimum volumes will be accounted for annually, that is to say, in the 12 months following the data of the agreement (between APMIC and the user) is addrounting be volumes (for example and as is hypothetical case; if the user's request for offer is accepted and it is agreed that the posting on June 1, 2024, the the accounting will be made until May 31, 2025). For this purpose, users must aubrint to the user's request for offer is accepted and it is agreed that the posting on June 1, 2024, the the accounting will be made until May 31, 2025). For this purpose, users must aubrint to the accounting will be accounting will be made until May 31, 2025). For this purpose, users must aubrint to the accounting the data of the accounting will be made until May 31, 2025). For this purpose, users must aubrint to the accounting will be accounting APUTC wil cany out a review every three (03) months of the imported volume. If its observed that in the first quarter at least 20% of the committed annual volume has not been met, the offer wil be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met, the offer wil be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met, the offer wil be cancelled; if it is observed that in the second quarter at least 45% of the committed annual volume has not been met, the offer wil be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the offer wil be cancelled.

APMIC wit carry a review every three (03) months of the volume of containers mobilized by the user who requested the temporary offer, it is determined during the review that the user will not be able to meet the required annual volume, then APMIC may cancel the promotion. If at least 20% of the minimum annual volume required is not needed in the first review, the temporary offer, it is determined during the review that the user will not be able to meet the required annual volume, then APMIC may cancel the promotion. If at least 20% of the minimum annual volume required is not needed in the first review, the temporary offer, it is determined during the review that the user will not be able to meet the accounting. It is useful to the minimum volume required for the application of this offer than not been reached. APMIC will drage the surface of a contractive and the review (12) months of accounting, it is writted that the minimum volume required for the application of this offer than not been reached. APMIC will drage the surface of a contractive and the review (12) months of accounting, it is writted that the minimum volume required for the application of this offer than not been reached. APMIC will drage the surface of a contractive and the review (12) months of accounting, it is writted that the minimum volume required for the application of this offer than not been reached. APMIC will drage the surface of a contractive and the review (12) months of accounting, it is writted that the minimum volume required for the application of this offer than not been reached. APMIC will drage the surface of a contractive and the tot of the tractive and the surface of the application of this offer than not been reached. APMIC will drage the surface of the application of the surface of the application of the application

The acce plance of an request for access to the offer will be subject to an evaluation of the projections and/or that the user has mobilized by the Port of Calao, in the year preceding the agreement to start accounting, at least 80% of the volume committed.

In the event that the offer has been cancelled to a user and the user requests access to a new offer, the rew access will be granted if the following condition is met that during the period of the cancelled offer (due to failure to comply with the volume commitment between the accounting start date and the cancellation date) it is verified that this user has handledifieded by IOT 3014 equal to or more than 80% of the total containers that have been directed through a temporary depost, (2014 or entraport) and unbaded/loaded by North Multi-purpose Temminal.

- This offer is subject to availability of snare determined by APMTC

#### 2.4 For import containers using APMTC as temporary depot (only for new users and no penalty)

s brokers, logistics operators, etc.) of containers with refrigerated cargo (reefer containers) that who address dry cargo to APMTC's Temporary Depot. Applies for new users

Target group: Importers of dry cargo containers (consignees or its representatives -customs brokers, logistics operators, etc.) of containers with refigerated cargo (revefic containers) that who address dry cargo to APMITC's Tem Temporary ofter: importers who register a quarterly movement of 60 containers (40 teet) of refigerated cargo will be granted a decount of USO 38.5 on the current prices, at the time of invicing, for the padage of services con and dry 6 of impection and monitoring (ubcetor in 15.2.4). ction 1.4.3.2), day 8 of operational area use (su section 1.4.3.3), day 8 of reefer power supply (sul ing of the integrated temporary storage service for import re - Effective: The offer may be requested from August 02, 2023 to june 30, 2024

#### - Require

um volumes will be counted in the 03 months following the date of the agreement (between APNITC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the posting on September 1, 2023, then the accounting will be made until November 30, 2023). The minim

This offer is not subject to an agre ment or mandatory commitment of movements by the user against APM Terminals Callao, which means that in case of non-compliance there will be no penalty. In the event that within the period of 3 months it is observed that the user does not reach the volume of the offer, then APMTC will cancel the offer to the user

## (Group 3) OFFER FOR PROMOTION FOR CONSOLIDATION/DESCONSOLIDATION OF CONTAINERS AND FOR USERS WHO HAVE ACCESSED THE PROMOTIONS OF GROUP 1 OF DT 3014

# Dear customers and users.

APM Terminals Calao (APMTC) has arranged offers for all those users who request a package of services and have requested and acce that the user would pay for the package of services consisting of any of the services specified in the Group 1 offer plus the consolidation sed the offers provided in Group 1, but referred specifically to the offers related to the use of the Temporary Dapot (code 3014); that is, it does not apply to the offers of Group 1 that refer to the use of APMITC as a port terminal. Offers refer to discounts (implying lower prices) on the agg acconsolidation service (in any of the forms specified in subsections 1.5.3.4 to 1.5.3.9 and 1.5.3.13 of the Tamiff List). The details of the offers of the package requested by the user are as detailed below. - Target group: Users who have agreements or have accessed offers (set out in group 1) regarding the use of APMTC DT 3014

#### - Temporary offer

lables of the Group 1 or Group 2 offers plus the consolidation/deconsolidation service of subsections 1.5.3.4 to 1.5.3.9 and 1.5.3.13 of the Tariff List. For example, if a user has acceded to numeral i of offer 1.1 of Group 1 offers (which grants a discount of USD 40 on the print additional) yeapest deconsolidation with a boldit of one of their containers, he would have to pay an additional USD 170 (gubeedicin 1.5.3.4 of the Tariff List) for a total of USD 433.1, however, with this offer and rank number of consolidation or deconvelvation non-animon 1 ... 1) Up to 99 operations: Discount of USD 30 off the global price that would be paid with any of the modalities o days free of storage), then he would have to assume a fee of USD 262.1 per 20-foot container and if he additio 408.1 per 20ft container (in the case of 40ft the fee with this promotion and example given will be USD 541.7)

iii) Between 100 to 289 operations: Discount of USD 50 off the global price that would be paid with any of the modalities of the Group 1 or Group 2 offers plus the consolidation service of subsections 15.3.4 to 15.3.9 and 15.3.13 of the Tariff List, For example, if a user has accoded to numeral if offer 1.1 of Group 1 offers (which grants a discount of USD 40 on the prices of subsections 16.1.1 or Group 2 offers plus the consolidation service of subsections 15.3.4 to 15.3.9 and 15.3.13 of the Tariff List for example, if a user has accoded to numeral if offer 1.1 of Group 2 offers plus the consolidation operations the user will have to pay an additional USD 176 (subsection 15.3.4 of the Tariff List for a subilit of Group 1 of Group 2 offers plus the consolidation operations the user will have to pay an additional USD 176 (subsection 15.3.4 of the Tariff List for a subili of USD 438.1; however, with this offer and rank number of consolidation operations the user will have to assume the of USD 438.1; however, with this offer and rank number of consolidation operations the user will have to assume of vectors of the consolidation operations the user will have to assume of vectors of the consolidation operations the user will have to assume of vectors of the consolidation operations the user will have to assume of vectors of the consolidation operations the vector will have to assume of vectors of the to assume of vectors of the consolidation operations the vector will have to assume of vectors of the vectors of the to assume of the vectors of the to assume of the total total

iii) Between 300 to 799 operations: Discourt of USD 70 of the global plote that would be paid with any of the modellies of the Group 1 or Group 2 offers plus the consolidation envice of subsections 1.5.3.4 to 1.5.3.9 and 1.5.3.13 of the Tariff List, For example, if a user has acceded to numeral iv of other 1.1 of Group 2 offers plus the consolidation operations to subsections 1.5.4. to 1.5.3.9 and 1.5.3.13 of the Tariff List, For example, if a user has acceded to numeral iv of other 1.1 of Group 1 of ther 1.1 of Group 2 offers plus the consolidation operations the user will have to pay an additional USD 176 (subsection 1.5.3.4 of the Tariff List) for a total of USD 423.1 however, with this offer and rank number of consolidation operations the user will have to assume parent USD 353.1 of the Tariff List for a total of USD 423.1 however, with this offer and rank number of consolidation operations the user will have to pay an additional USD 176 (subsection 1.5.3.4 of the Tariff List) for a total of USD 423.1 however, with this offer and rank number of consolidation operations the user will have to assume parent USD 353.1 however, with this offer and rank number of consolidation operations the user will have to pay an additional USD 176 (subsection 1.5.3.4 of the Tariff List) for a total of USD 423.1 however, with this offer and rank number of consolidation operations the user will have to assume parent USD 353.1 however, with this offer and rank number of consolidation operations the user will have to assume parent bits user will have to assume parent USD 353.1 however, with this offer and rank number of consolidation operations the user will have to assume parent USD 353.1 however, with this offer and rank number of the 1.5.1 however, with this offer and rank number of consolidation operations the user will have to assume parent bits assume pare

(v) 2000 operations: Discount of USD 110 of the global price that would be paid with any of the modalities of the Group Tor Group 2 offers plus the consolidation decompositiation service of subsections 15.3.4 to 15.3.9 and 15.3.13 of the Tariff List. For example, if a user has acceled to numeral iv of offer 1.1 of Group 1 offers (which grants a discount of USD 20 on the prices of subsections 15.3.4 to 15.3.9 and 15.3.13 of the Tariff List. For example, if a user has acceled to numeral iv of offer 1.1 of Group 1 offers (which grants a discount of USD 20 on the prices of subsections 14.1.1 or 14.1.2 and 30 days free of storage). The have to assume a few of USD 22.7 per 2-N-do container and if haddinosity request discountidation of the consolidation of example plants and thave to assume a payment of USD 31.8 it per 20 container (in the see with fits prices and the list of the few with fits prices and with use USD 31.8 it per 20 container (in the see with fits prices and with use USD 31.8 it per 20 container (in the see with fits prices and with user USD 31.8 it per 20 container (in the see with fits prices and with user USD 31.8 it per 20 container (in the see with fits prices and with user USD 31.8 it per 20 container (in the see with fits prices and with user USD 31.8 it per 20 container (in the see with fits prices and with user USD 31.8 it per 20 container (in the see with fits prices and with user USD 31.8 it per 20 container (in the see with fits prices and with user USD 31.8 it per 20 container (in the set with fits prices and with user USD 31.8 it per 20 container (in the set with fits prices and with user USD 31.8 it per 20 container (in the set with fits prices and with user (in the set with fits prices and user and the set with fits prices an

- Effective: The offers i), ii) y ii) is valid from August 02, 2023 to June 30, 2024. Offer iv) will be valid from March 22, 2024 to June 30, 2024

ation/deconsolidation service in any of its forms) will be granted an automatic discount of USD 30 regardless of the number of consolid ted the service package (composed of group 1 offers plus the consolida tion operations. In case users want to access the discounts of offers ii and iii, they will have to apply to APMTC and oo

The number of deconsolidation consolidation consolidations of promotions i, iii y i will be counted considering the accounting period established in groups 1 and 2 of the offers. For example, if a user has an agreement, in group 1 or 2 of offers, whose accounting is annual and it has started on April 1, 2023 and ends on March 30, 2024, then the number of consolidation operations will be accounted for within the last period described. Another example, if the agreement, for group 1 or 2 of offers was semiannual accounting and has a start date of September 1, 2023 and an end date of February 28, 2024, then the accounting of the number of perations will be carried out within of this last period described.

APIC of any out a view every hree (03) months of the volume of containers moved by the user who equested emporary offers i and i; if it is writed, outry out a view every three (03) months of the volume of containers moved by the user who equested emporary offers i and i; if it is writed, outry on the able to match the required annual volume, then APIC may cannot be offers. In its regard, it at least 15% of the minimizantial offers in the required annual volume, then APIC may cannot be offers. In its regard, it at least 15% of the minimizantial volume required annual volume, then about the required annual volume required and reached in the three required annual volume required and reached in the three required annual volume required and reached in the three required annual volume required and reached in the three required annual volume required and reached in the three required annual volume required and reached in the three required annual volume required and reached in the three reached in the three reached. It is annual reached in the three reaches in the reached in the three reaches in the reached in the three reaches. Integrating the reaches in the reach

APMIC will conduct a review every three (03) months of the volume of containers mobilized by the user requesting the temporary offer (in it is verified, during the review, that the User does not meet or will be able to meet the required annual volume, then APMIC on y cancel the promotion or indicate to the user that the previous temporary offer (in which could be offer iii or ii) will apply. In this regard, if at least 25% of the minimum annual volume required in on term risk will be canceled to reliaining this, the descont from offer iii will be applied to the operations of the string during the temporary offer (in will be canceled or, talling this, the descont from offer iii will be applied to the operations of the string during the temporary offer (in will be canceled or, talling this, the descont from offer iii will be applied to the operations of the string during the temporary offer (in will be canceled or, talling this, the descont from offer iii will be applied to the operations of the string during the temporary offer (in will be canceled or, talling this, the descont from offer iii will be applied to the operations of the string during the temporary offer (in will be canceled or, talling this, the descont from offer iii will be applied to the operations of the string during the temporary offer (in will be canceled or, talling this, the descont from offer iii will be applied to the operations of the string during the temporary offer (in will be canceled or, talling this, the descont from offer iii will be applied to the operations of the string during during during the string during duri This offer will be subject to space availability confirmed by APMTC

#### (Group 4) OFFER FOR APPLICATION OF SERVICE PACKAGES TO CONTAINERS IN APM TERMINALS CALLAO - SECOND MODALITY

ers and users

APM Terminals Calao (APMTC) has set special offers for all those dry import containers that are served through the Port Terminal or through its Temporary Decot (DT 3014) for which service packages are requested. The offers refer to discounts (which imply lower prices) and apply by package type which are detailed below

- Target group: Users who do not have agreements or have not accessed DT 3014 offers (Groups 1 or 2)

# Temporary offer i): without the need to reach a certain volume

Group of packets 1- DT 3014: integrated service for 20 or 40 foot import dry container (subsections 1.4.1.1 and 1.4.12 of the Tariff List), and container deconsolidation in any of its forms (subsections 1.5.3.4, 1.5.3.5 and 1.5.3.5 of the Tariff List). For packages that are claimed, a global discount of USD 30 for a 20-foot co

Group of packets 2 - 07 3014. Integrated service for 20 or 40 foot import dry container (subsections 1.4.1.1 and 1.4.1.2 of the Tariff List), container deconsolidation in any of its forms (subsections 1.5.3.4, 1.5.3.5 and 1.5.3.6 of the Tariff List) and use of area from day 11 to day 20 (subsection 1.4.1.3 of the Tariff List). For packages that are claimed, aglobal discount of USD 183.1 for a 20-box container and USD 371.7

Group of packets 3 - DT 3014: integrated service for 20 or 40 foot import dry container (subsections 1.4.1.1 and 1.4.1.2 of the Tariff List) and partial consolidat ildation of container with dry cargo (subsection 1.5.3.13 of the Tariff List). For packages that are claimed, a discount of USD 25 for a 20-foot co Group of packets 4 - as port terminal: cargo portion of Standard Service for 20 or 40 foot import dy container (subsections 1.1.2.1 and 1.1.2.2 of the Tariff Liat) and container deconsolidation in any of its forms (subsections 1.5.3.4, 1.5.3.5 and 1.5.3.5 of the Tariff Liat). For packages that are claimed, a global discount of USD 29.6 for a 24-foot container and USD 39.6 for a 40-foot container deconsolidation in any of its forms (subsections 1.5.3.4, 1.5.3.5 and 1.5.3.5 of the Tariff Liat).

Group of packets 5 - as port terminal: cargo portion of Standard Service for 20 or 40 foot import dry container (subsections 1.1.2.1 and 1.1.2.2 of the Tariff List) and partial consolidation of container with dry cargo (subsection 1.5.3.13 of the Tariff List). For packages that are daimed, a global discount of USD 13.6 for a 20 or 40 foot container will be granted

offer ii): based on demand or volume: for users who carry out more than 200 deconsolidation operations in an annual period

Group of packets 2- 07 3914: Integrated service for 20 or 40 foot import dy container (subsections 14.1.1 and 14.1.2 of the Tairff Lid), container deconsolidation in any of to forms (subsections 15.3.4, 15.3.5 and 15.3.6 of the Tairff Lid), and use of area from day 11 to day 20 (subsection 14.1.3 of the Tairff Lid), container deconsolidation in any of to forms (subsections 15.3.4, 15.3.5 and 15.3.6 of the Tairff Lid), and use of area from day 11 to day 20 (subsection 14.1.3 of the Tairff Lid), container deconsolidation in any of to forms (subsections 15.3.4, 15.3.5 and 15.3.6 of the Tairff Lid), container deconsolidation in any of to forms (subsections 15.3.4, 15.3.5 and 15.3.6 of the Tairff Lid), container deconsolidation in any of to forms (subsections 15.3.4, 15.3.5 and 15.3.6 of the Tairff Lid), container deconsolidation in any of to forms (subsections 15.3.4, 15.3.5 and 15.3.6 of the Tairff Lid), container deconsolidation in any of to forms (subsections 15.3.4, 15.3.5 and 15.3.6 of the Tairff Lid), container deconsolidation in any of to forms (subsections 15.3.4, 15.3.5 and 15.3.6 of the Tairff Lid).

The number of operations of the temporary offer i will be counted within the annual period agreed between APMIC and the user, the earliest date to start the accounting will take place until June 30, 2024. As a second hy example, if the user's offer request is accepted and it is agreed that the accounting will bean on September 1, 2023, then the accounting will take place until June 30, 2024. As a second hy example, if the user's offer request is accepted and it is agreed that the accounting will bean on September 1, 2023, then the accounting will take place until June 30, 2024. As a second hy example, if the user's offer request is accepted and it is agreed that the accounting will bean on September 1, 2023, then the accounting will take place until June 30, 2024. As a second hy accepted accepted and it is agreed that the accounting will bean oned by the user who negoted the first. This required annual volume, then the required annual volume, then the required annual volume, the other is accepted in the encoder in the first revision, temporary offer i will be canceled offs of the minimum annual volume equired is not reached in the first revision, temporary offer i will be canceled.

In case of changes in the prices of the Tariff List subsections involved in these offers (1.1.2.1, 1.1.2.1, 1.4.1.2, 1.4.1.3, 1.5.3.4, 1.5.3.5, 1.5.3.6 y 1.5.3.13), the discounts described apply to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tariffs and Commercial Policy of APM Terminals Calao

This offer will be subject to space availability confirmed by APMTC

#### (Group 5) OFFER FOR UN 3077 CARGO CONTAINERS CLASSIFIED AS IMO 9 IN APM TERMINALS CALLAO'S TEMPORARY DEPOT

4PM Terminals Callao (APMTC) has set that export containers with dangerous cargo UN 3077 (UN 3077) - such as zinc oxide, abletic acid and others - dessified as IMO 9 class, to be serviced through its Temporary Depot (code 3014), the price of the special service "Treatment of dangerous cargo IMO 9" will not be applied; in other words, a 100% discourt will be applied to the price of subsection 1.5.1.3. This promotion will be trom January 1, 2021 to June 30, 2024.

# (Group 6) SPECIAL OFFER FOR FISHMEAL IN CONTAINERS

## ar customers and users

Dear customers and users

APM Terminals Callao (APMTC) has set offers for containers with fishmeal. These offers are governed under the following terms:

6.1 First promotion - as port terminal: exemption from the price of the special service "high cube containers"

Any user who loading fishmeal in containers will not be charged the price of the special service "high cube containers" of subsection 1.5.1.8. This offer will be valid from July 1, 2021 to June 30, 2024

### 6.2 Second promotion - as port terminal: discount on standard service tariff and on special service price of special treatment of Hazardous cargo

- Target group: Exporters (forwarders or its representatives -customs brokers, logistics operators, etc- and other users) of of 40-foot containers with fishmeal

. Temporary offer: Fishmeal exporters shall be granted discounts on the standard cargo-portion service tariffs actually charged (subsection 1.1.2.1 of the Tariff List) and on the price of the special service for special treatment of hazardous cargo IMO 9 (subsection 1.5.1.2 or 1.5.1.3 of the Tariff List)): The details of the discounts are as hollows:

(i) For any user who loading fishmeal: A discount of USD 96.7 will be granted on the general promotion tariff of standard cargo-portion service applicable to full 40-foot co the Tariff List) shall be exempted. Includes exemption from payment of the price for the special high-cube service (subsection 1.5.1.8). tainers (subsection 1.1.2.2 of the Tariff List) that is in force at the time to request si reatment special for haz ardous goods (subsection 1.5.1.2 or 1.5.1.3 of

- Effective: The offer is valid from November 15, 2023 to April 30, 2024 and from May 2, 2024 to August 30, 2024.

- This offer will be subject to space availability or capacity confirmed by APMTC

#### 6.3 Third pro ion - as temporary depot: discount on the price of the integrated temporary depot service and on the special service for treatment of Hazardous cargo cor

- Target group: Exporters (forwarders or its representatives -customs brokers, logistics operators, etc- and other users) of of 40-foot containers with fishmeal that direct their containers to DT 3014.

- Temporary offer: "Fahmeal exporters shall be granted discourts on the prices of the integrated temporary depot service (subsection 1.5.1.2 or 1.5.1.3 of the Tariff List) and an extension of days free of use of the operational area; Includes exemption from payment of the special temporary depot service for special temporary depot service (subsection 1.5.1.2 or 1.5.1.3 of the Tariff List) and an extension of days free of use of the operational area; Includes exemption from payment of the special temporary depot service for special temporary depot service for special temporary depot service (subsection 1.5.1.2 or 1.5.1.3 of the Tariff List). It includes exemption from payment of the price of the special temporary depot service for 40-foot dry containers and a total of 14 days of use of the operational area; includes exemption from payment of the special temporary depot service for 40-foot dry containers and a total of 14 days of use of the operational area; includes exemption from payment of the special temporary depot service for 50-foot dry containers and a total of 14 days of use of the operational area; includes exemption from payment of the special temporary depot service for 50-foot dry containers and a total of 14 days of use of the operational area; includes exemption from payment of the price of the special temporary depot service for 50-foot dry containers and a total of 14 days of use of the operational area; includes exemption from payment of the price of the special temporary depot service for 50-foot dry containers and a total of 14 days of use of the operational area; includes exemption from payment of the price of the special temporary depot service for 50-foot dry containers and a total of 14 days of use of the operational area; includes exemption from payment of the price of the special temporary depot service for 50-foot dry containers and a total of 14 days of use of the operational are - Effective: The offer is valid from November 15, 2023 to April 30, 2024 and from May 2, 2024 to August 30, 2024.

- This offer will be subject to space availability or capacity confirmed by APMTC

## (Group 7) OFFER FOR MINERAL LOADED IN CONTAINERS

# Dear customers and users,

APM Terminals Callao (APMTC) has set offers for loading containers with mineral concentrates (bulk) and/or minerals in big bags. These offers are governed by the following terms:

### 7.1 First offer - as port terminal: discount on standard service and special service of treatment of hazardous cargo

- Target group: Exporters (forwarders or its representatives -customs brokers, logistics operators, etc- and other users) of of 20-foot containers with mineral concentrates and/or minerals in big bags

- Temporary offer: Exporters of minerals on minerals and/or minerals in big bags shall be granted discounts on the standard service portion charge actually charged (subsection 1.1.2.1 of the Tariff List) and on the price of the special special handling service for hazardous cargo IMO 9 (subsection 1.5.1.3 of the Tariff List). On the one hand, the discounts will apply to all exporters of mineral but on the ther hand, other discounts will apply subject to a certain volume of containers; the delais of the discounts are as follows:

(i) Any user who loading minerals and/or minerals in big bags. A discount of USD 34.6 will be granted on the general promotional lariff of standard service-land portion applicable to full 20-foot containers (subsection 1.1.2.1 of the Tariff List) in effect at the time of nequesting the standard service. Likewise, the payment of the special service for special treatment of dangerous and of the applicable to full 20-foot containers (subsection 1.1.2.1 of the Tariff List) in effect at the time of nequesting the standard service. Likewise, the payment of the special service for special treatment of dangerous and of the applicable to full 20-foot containers (subsection 1.1.2.1 of the Tariff List).

(ii) Users who bading equal to or more than 5 thousand containers of mineral concentrates and/or minerals in big bags in an annual period: A discount of USD 67.6 will be granted on the general promotional rate of standard service-land portion applicable to ful 20-boot containers (subsection 1.1.2.1 of the Tariff List) in effect at the time of requesting the standard service. Likewise, the payment of the special service for special treatment of dangerous cargo will be exempt (subsection 1.1.2.1 of the Tariff List) in effect at the time of requesting the standard service. Likewise, the payment of the special service for special treatment of dangerous cargo will be exempt (subsection 1.1.2.1 of the Tariff List).

Effective: The offer i) is valid from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2023 to March 31, 2024 to June 30, 2024; while, the offer ii) may be requested from September 08, 2024 to March 31, 2024 to March

The volume of containers of the temporary offer i will be counted within the annual period agreed by APAITC and the user (for example and as a hypothetical case; if the user's nequest for offer is accepted and it is agreed that the accounting will start on September 10, 2022, then the accounting will be carried out until September 00, 2024) Regarding the temporary offer i, APAITC will carry out a review every three (03) months of the economited annual volume. It is absented that in the first quarter tale tast 17. Si of the committed annual volume has not been made, the start and the temporary offer i, APAITC will carry out a review every three (03) months of the economited annual volume. This observed that at least 40% of the committed annual volume has not been made, the start and the temporary offer i, APAITC will carry out a review every three (03) months of the economited annual volume has not been made, the start of the temporary offer i, APAITC will carry out a review every three (03) months of the economited annual volume has not been made, the review of the committed annual volume has not been made, the review of the review every the review every three (03) months of the economited annual volume has not been made, the review of the review every temporary of temporary o

- This offer will be subject to space availability confirmed by APMTC

# 7.2 Second offer - as DT 3014: discount on standard service, on special service of treatment of hazardous cargo and 14 days use of operating area

- Taroet group: Exporters (forwarders or its representatives -customs brokers, logistics operators, etc- and other users) of of 20-foot containers with mineral concentrates and minerals in big basis that direct their containers to DT 3014.

- Temporary offer: Exporters of mineral concentrates and minerals and big bags will be granted discounts on the prices of the integrated service of special trainfil List), on the price of the special service of special treatment of hazardous cargo MO9 (subsection 1.5.1.3 of the Tariff List) and an extension of days free of use of the operating area. Specifically, a price of USD 190 per 20-hoot container will be changed which induces the integrated temporary depot service for 20-hoot days free of use of the operating area. This change is the changed which induces the integrated temporary depot service for 20-hoot day containers and a lotal of 14 days of use of the operating area. Special Treatment Service - Effective: The promotion will be valid from May 2. 2024 to June 30 2024

- This offer will be subject to space availability confirmed by APMTC

#### (Group 8) OFFER FOR TRANSSHIPMENT OF CONTAINERS

# Dear customers and users

#### APM Terminals Callao (APMTC) has set offers for transshipment conta iners that are loaded/unloaded by the North Multi-ourcose Terminal. These offers are governed by the following terms

8.1 First offer - 800 containers: discount on standard service, 14 days free of use of operational area (possible extension) and exemption from change of status

# - Target group : Users (shipping lines or its representatives) that move transhipment containers through the North Multi-purpose Terminal

- Temporary offer: Users who move transshipment containers equal to or greater than 800 weekly average containers, over a period of one year, will be granted a discount on the standard container trans ant service tariff and an extension of free days, as follows

er. Usen swin move transshorment containers equal to gradest han 8UU weely average containers, over a penci of one year, will be granted a discount on the standard container transshorment source than dia netension of the disp, as tolows: (1) ≥ 800 average containers per week = They will be charged a standard service tariff of USD to for 2-box transshorment containers (subsection 1.1.2.8 of the Tariff List). It includes for both sizes of containers (20 and 40 feet) the granting of 14 free days (which represents 12 dational feet dational feet

(ii) Likewise, users who comply with the weekly average movement of 800 containers, during an annual period, will be exempted from the payment of the container status change surcharge

- Effective: The offer may be requested from November 15, 2023 to January 31, 2024

The average weekly volume shall be counted and calculated within the annual beniced acreed between APMTC and the user for example and as a hypothetical case: if the user's request for offer is accepted and it is acreed that the accountion will start on December 1 2023, then the accounting will be carried out until November 30, 2024i. All hopes of transhoment (full cycle or inte The amongs prevery fouries and contract to contract to contract and and contract to contra

APMTC will carry out a review every three (03) months of the weekly average of the transshipment containers mobilized by the user who requested the offer, If it is verified, during the review, that the user has not complied with the required weekly average, then APMTC may cancel the offe

# - This offer will be subject to space availability confirmed by APMTC

8.2 Second offer - 1,800 containers: two types of discounts on standard service, 14 days free of use of operating area, free pool and exemption from change of status

- Target group : Users (shipping lines or its representatives) that move transhipment containers through the North Multi-purpose Terminal

- Temporary offer: Users who move transshipment containers equal to or greater than 1,800 weekly average containers, over a period of one year, will be granted similar discounts or benefits as offer 9.1 but an additional discount is added to the standard service tariff that will be related in discount amounts and other additional benefits. The details of the discounts and benefits of this offer are as follows i) Charge of an initial offer tariff of USD 90 for standard 20-foot transhipment container service (Section 1.1.2.7 of the Tariff List) and USD 120 for standard 40-foot transhipment container service (Section 1.1.2.8 of the Tariff List).

i) Additional discount of USD 35 per container on the offer tarffits described in item i), but which may be given for a maximum of 17,715 containers in each quarter during the annual period of the offer. This additional discount, which will be applied for a period of one year, will be for one time only, that is, it will not apply in case of renewal of this offer ii) Grantino of 14 free days of use of the operational area (which represents 12 additional free days to the 48 free hours currently granted) for both container sizes (20 and 40 feet).

# iv) Granting of a free pool of 300 TEU's (use of operational area of 300 TEU's of transhipment without time limit). This free pool is granted to those containers that have exceeded the 14 days of use of the operational area.

v) Exemption from payment of container status change surcharge

- Effective: The offer may be requested from April 11, 2023 to December 31, 2023

The average weekly volume will be counted and calculated within the annual period agreed by APM/CD and the user, which may be the earliest date to start counting on April 1, 2023 (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the annual period agreed by APM/CD and the user, which may be the earlied out until March 31, 2024. As a second hypothetical case; if the user's request for offer is accepted and it is agreed that the annual period agreed by APM/CD and the user, which may be the earlied out until March 31, 2024. As a second diverse is accepted and it is agreed that the accounting will be carried out until March 31, 2024. As a second diverse, if the user's offer request is accepted and it is agreed that the accounting will be carried out until March 31, 2024. As a second diverse is accepted and it is agreed that the accounting will be carried out until March 31, 2024. As a second diverse is accepted and it is agreed that the accounting will be carried out until 31 May 2024. All types of transforment (full cycle or interferminal) shall be accounted. Accounting is a period that the second more a weekly average of 1,000 containers (which results in a weekly average of 3,000 movements; 3,400 full cycle containers and 100 inter-terminal movements). The average weekly count is calculated by taking into account the total number of transforment containers moved during this annual period and is divided by the number of weeks in that period (which is 52 weeks). For this purpose, users should submit to the APMITC commercial area the projections of transhipment containers that they will be mobilize during the annual period indicated above, in order to determine the availability of terminal space. APMITC will cany out a review every three (03) months of the weekly average of the transshipment containers mobilized by the user requesting the offer; if it is verified, during the review, that the user has not met the required weekly average, then APMITC may cancel the offer.

APMITC will carry out a review every three (03) months of the weakly average of the Itanshipment containers mobilized by the user requesting the offer; if it is verified, during the review, that the user is meeting the required weakly average. The APMITC will apply for that quarterly period and with respect to the item i discount, the additional discount of USD 35 (on the initial promotional lariffs for standard service of containers of item i) over a maximum of 17,715 transhipment containers. This quarterly discount, which will apply to a maximum of 17,715 transhipment containers. This quarterly discount, which will apply to a maximum of 17,715 transhipment containers. This quarterly discount, which will apply to a maximum of 17,715 transhipment containers. This quarterly discount, which will apply to a maximum of 17,715 transhipment containers. This quarterly discount, which will apply to a maximum of 17,715 transhipment containers. This quarterly discount, which will apply to a maximum of 17,715 transhipment containers. This quarterly discount with respect to the item i discount, the additional discount of 17,715 transhipment containers. This quarterly discount has been average. The territ discount and the free pool of 300 TEU, the user will have to assume the payment of the territ applicable from the 15th day onwards (subsections 1.2.6.4 and 1.2.7.4 of the Tariff List). However, if the user mobilizes more than 1,800 containers per weak on average. APMITC may extend, subject to space availability, the days of use of the correlisting and the territ discount dis discount discount discount discount discount discount discou

operational area for those containers that remain more than 14 days and cannot access the free pool. The extension days would be up to seven days at most. If APMTC grants this benefit to a user, it will be applied to other users under similar conditions.

## This offer will be subject to space availability confirmed by APMTC

## 8.3 Third offer - 1,300 containers: two types of discounts on standard service, 14 days free of use of operating area, free pool and exemption from change of status

- Target group : Users (shipping lines or its representatives) that move transhipment containers through the North Multi-purpose Termina

rary offer. Uses who move transhipment containers equal to or greater than 1,300 weekly average containers, over a period of 2 years and 3 months (09 quarters or 27 months in total), will be greated similar discounts and benefits as offer 9.1 but an additional discount is addeed to the standard service tariff that will be reflected in discounts and benefits. The details of the discounts and therefore that will be greated service tariff that will be reflected in discount and other additional discount is addeed to the standard service tariff that will be reflected in discounts and other additional discount is addeed to the standard service tariff that will be reflected in discount and other additional discount is addeed to the standard service tariff that will be reflected in discount and other additional discount is addeed to the standard service tariff that will be reflected in discount and other additional discount is addeed to the standard service tariff that will be reflected in discount and other additional discount and other additional discount is addeed to the standard service tariff that will be reflected in discount and other additional discount is addeed to the standard service tariff that will be reflected in discount and other additional discount is addeed to the standard service tariff that will be reflected in discount and other additional discount and other additionad discount and other additionad discount an ) Charge of an initial offer tariff of USD 90 for standard 20-toot transhipment container service (Section 1.1.2.7 of the Tariff List) and USD 120 for standard 40-foot transhipment container service (Section 1.1.2.8 of the Tariff List).

i) Additional discount of USD 35 per container on the offer tariffs described in item i), but which may be given for a maximum of 9,143 containers in each quarter during the 09-quarter (or 27-month) period of the offer. This additional discount, wich will appled during the period of the offer.

ii) Granting of 14 free days of use of the operational area (which represents 12 additional free days to the 48 free hours currently granted) for both container sizes (20 and 40 feet).

iv) Granting of a free pool of 300 TEU's (use of operational area of 300 TEU's of transhipment without time limit). This free pool is granted to those containers that have exceeded the 14 days of use of the operational area.

v) Exemption from payment of container status change surcharge.

vi) Additional discount of USD 25 per container on the promotional tariffs described in item (i), but which may be given for a maximum of 3,976 containers and only in the first quarter of the offer. This discount does not apply for the remaining eight (08) quarters and will not apply in the event of renewal of the offer

The average weekly volume will be counted and calculated within the period of 2 years and 3 months agreed by APMITC and the user, which may be the earliest date to start counting on October 1, 2023. (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the accounting will be carried out until December 31, 222.5. As a second hypothetical example. The user's offer request is accepted and it is agreed to stat accounting on February 01, 2024, have a counter of a second grave and the agreed to stat accounting period the user of the request is accepted and it is agreed to stat accounting period the user of the request is accepted and it is agreed to stat accounting period. Building the accounting period the user of the request is accepted and it is agreed to stat accounting period. Building the accounting period the user of the request is accepted and it is agreed to stat accounting period. Building the accounting period the user of the request is accepted and it is agreed to stat accounting period. Building the accounting period the user of the request is accepted and the period and divided by the number of tweels in that period (which is 117 weeks). For this purpose, users doubt addiver to the APPIIC Comments that they all building the accepted addiver, in order to be applic formating and the period accepted addiver, in order to be applic formating and the period. Accepted and the period accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be applic formating and the period. Accepted addiver, in order to be appliced formating and the period. Accepted addiver, in

APMITC wil carry out a review every three (IQ) months of the weekly average of the transitionment containers mobilized by the user requising the office, if it is verified, during the review, that the user has been complying with, at least 90% of the required weekly average, then APMITC will apply for that quarterly period and with respect to the item i discount, the additional discount of USD 35 (on the initial promotional lattifts for standard service of containers of item i) over a maximum of 9,143 containers, will be granted in the form i d exet(in the or additing agreed upon by the shipping line and APMITC.

The additional discount of USD 25 per container set out in item vi) will be granted as long as the user has complied, during the first quarter, with the committed weekly average (that is, with 1,300 weekly average containers). This discount will only be granted for the first quarter of the offer.

In case of containers that exceed the 14 days fire of use of the operating area and the free good of 300 TEU, the user will have to assume the payment of the tariff applicable from the 15th day onwards (subsections 12.6.4 and 12.7.4 of the Tariff List), However, if the user mobilizes more than 1,000 containers per week on average, APMIT may extend, subject to space availability, the days of use of the operational area for those containers that remain more than 14 days and cannot access the free good. The extension days would be up to seven days at most. If APMITC grants this benefit to a user, it will be applied to other users under similar conditions.

# This offer will be subject to space availability confirmed by APMTC

8.4 Fourth offer: discounts and extension of free days per vessel arrived

- Target group : Users (shipping lines or its representatives) that move transhipment containers through the North Multi-purpose Terminal

Temporary offer: Users who move transshipment or ntainers equal to or greater than 250 and 430 containers per vessel be granted the discounts and benefits described below

i) 250 to 419 transhipment containers for each arrival: a 50%, discount on the operating area use price will be granted for the period from day 3 to day 14. This discount is for each veesel that arrives at the North Multipurpose Terminal and comples with the mentioned volume

ii) > 430 containers: Charging of a promotional tariff of USD 90 per standard 20-koot transshipment container service (Section 1.1.2.7 of the Tariff List) and USD 120 per standard 40-koot transshipment container service (Section 1.1.2.8 of the Tariff List), as well as 14 days free of use of the operational area

- Effective: The offer is valid to vessels arriving between January 26, 2024 and February 29, 2024. ments: Users who apply for promotions 8.1, 8.2 and 8.3 will not be able to apply for this offer

- This offer will be subject to space availability confirmed by APMTC

#### (Group 9) OFFER FOR GASIFICATION OF FULL CONTAINERS

# ar customers and users,

APM Terminals Calao (APMTC) has set a offer for volume for those users (shipping lines) that demand the gasification service of full containers that requires the joint supply of N2 and CO2 as a port terminal. This offer is governed by the following terms:

(1) = 150 containers quartery - Tarff of USO 450 which will include the following: combined guartations on Quarter of the openating area in addition to the 72 hours guarted as part of the standard shipping containers envice (up to a total of 5 fee days of use of the openating area as a PPI Tarman).

Effective: The offer may be requested from July 01, 2023 to March 31, 2024 and from May 2, 2024 to June 30, 2024.

### - Requeriments

m volumes of promotions will be counted on a quarterly basis; that is, in the 03 months following the date of the agreement (between APVITC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for promotion is accepted and it is agreed that the accounting on April 1, 2024, then the accounting will be carried out unit June 30, 2024).

If, after the three (03) months of accounting, is verified that the minimum volumes required for the acolication of these offers have not been reached. APMIC will charoe the maximum tariff in subsection 12.13.1 in effect at the time of verification and the charoe of the charoe of status sucharoe (subsection 9.3.1.1 of the Tariff List) in effect at the time of biling will apoly

- This offer will be subject to space availability confirmed by APMTC

#### (Group 10) OFFER FOR EMPTY EXPORT CONTAINERS

## Dear customers and users,

APM Terminals Callao (APMTC) has set an extension of free hours of stay for those empty containers that are loaded (exported) by the North Multipupose Terminal. This offer is governed by the following terms:

- Target group: Users (shipping lines or their representatives) that move empty export containers through the Multipurpose North Termina

Temporary offer. Shipping lines that move, during an annual period, a volume equal to or greater than 25,000 empty export containers will be granted the right to request, for each vessel, an extension of 24 hours of fee time that are additional to the hours of use of operating area free set out in subsection 12.5.1 of the Tariff List (which are 72 hours of free time), which means that shipping lines that move, during an annual period, a volume equal to or greater than 25,000 empty export containers will be granted the right to request, for each vessel, an extension of 24 hours of free time that are additional to the hours of use of operating area free set out in subsection 12.5.1 of the Tariff List (which are 72 hours of free time), which means that shipping lines that move, during a free time that are additional to the hours of use of operating area free set out in subsection 12.5.1 of the Tariff List (which are 72 hours of free time), which means that shipping lines that means that are additional to the hours of use of operating area free set out in subsection 12.5.1 of the Tariff List (which are 72 hours of free time), which means that shipping lines that means that are additional to the hours of use of operating area free set out in subsection 12.5.1 of the Tariff List (which are 72 hours of free time), which means that shipping lines that means that are additional to the hours of use of operating area. The detail of the offer is as follows:

(i) Movement of 500 empty export containers per vessel = Empty containers on this vessel will be granted a total of 96 hours of free operating area use (which represents an extension of 24 hours free from the 72 hours free granted under subsection 1.2.5.1 of the Tariff List).

(ii) Movement of 1.000 emoty export containers per vessel = Emoty containers on this vessel will be granted a total of 120 hours of free goerating area use (which represents an extension of 48 hours free from the 72 hours free granted under subsection 1.2.5.1 of the Tariff List).

## Effective: The offer is valid from January 03, 2024 to December 31, 2024.

- Requerim

# The fee fee of 96 or 120 hours of 96 or 120 hours of operational area use will have b be requested by each vessel had docks. In those cases where the shipping line dd not request the application of the offer to a particular vessel, then that vessel will only be granted the fee time set out in subsection 12.5.1 of the Tarif List (that is, 72 hours fee of use of the operating area), danging for use of the operating area from each offer to a particular vessel, then that vessel will only be granted the fee time set out in subsection 12.5.1 of the Tarif List (that is, 72 hours fee of use of the operating area), danging for use of the operating area from each offer to a particular vessel.

#### The minimum volume of 25 thousa iners will be accounted during the period of validity of the offer; that is, between January 03, 2024 until December 31, 2024.

If, during weekly or monthly reviews of the volume of empty containers exported by shipping lines, it is found that the shipping line will not reach the minimum volume required during the annual period, then the offer to that shipping line will be cano - This offer will be subject to space availability confirmed by APMTC

(Group 11) SPECIAL OFFER FOR BREAK BULK CARGO - TEMPORARY DEPOSIT OF APM TERMINALS CALLAO

# Dear customers and users

APM Terminals Calan (APMTC) has set a series of senial offers for the special service "Internated Termonary Denot Service" for break hulk canon. It should be noted that each of the special offers is independent (if is not cumulative) for example if the user chooses common 11 4 he would not he able to annot he many to normation 11 11 2 or 113. Special offers are novement by the following terms

#### 11.1 Extension of free days applicable to break bulk cargo in general and to subsection 2.4.1.1 of the Tariff List

- Target group: Users (importersionsignees and Exportersionwarders or its representatives - customs brokers, logistics operators, etc.) of break buik cargo in general and that demand the integrated service of temporary storage with 10 days of use of the operational area (subsection 2.4.1.1 of the Tariff List)

- Segment: Users entering large volumes break bulk cargo at APM Terminals Callag Temporary Depot (DT 3014)

- Temporary offer. Users who guarantee minimum amounts of more than 2,500 and 3,500 tonnes using the APMITC's Temporary Depot will be granted an extension of the operating area use (taking into account the 10 days including in the Integrated Service of Temporary Depot sections 2.4.1.1 of the Tariff List), accounting to the table below

- (i) ≥ 2,500 to 3,499 tonnes = 5 additional days off (on top of the 10 already granted in subsection 2.4.1.1 of the Tariff List)
- (i) ≥ 3,500 tonnes = 10 additional days off (on top of the 10 already granted in subsection 2.4.1.1 of the Tariff List)

- Effective: Users may apply or request this promotion from August 02, 2023 until June 30, 2024.

- Requeriments: The minimum volume of 2,500 or 3,500 tonres will be accounted for annually, that is within twelve (12) months from the date of the agreement (between APMICC and the user) to commence the accounting of volumes (for example and as a hypothetical case; if the user's promotion request is accepted and ii is agreed that the accounting will commence on September 1, 2023, then the accounting user) to out mill August 31, 2020, for this purpose, users should submit to APMICS commencial area the projections of tonnes they will import during the projection indicated adove, in order to determine the availability of farminal space. If, after the twelve (12) months of accounting, if is verified that the minimum volumes required for the application of these offers have not been reached, APMIC will charge provide the availability of farminal space. If, after the twelve (12) months of accounting, if is verified that the minimum volumes required for the application of these offers have not been reached, APMIC will charge provide the availability of farminal space. If, after the twelve (12) months of accounting, if is verified that the minimum volumes required for the application of these offers have not been reached, APMIC will charge provide the availability of farminal space. If, after the twelve (12) months of accounting, if is verified that the minimum volumes required for the application of these offers have not been reached, APMIC will charge provide that the twelve (12) months of accounting, if is verified that the minimum volumes required for the application of these offers have not been reached, APMIC will charge provide that the twelve (12) months of accounting, if is verified that the minimum volumes required for the application of these offers have not been reached, APMIC will charge provide that the twelve (12) months of accounting, if is verified that the minimum volumes required for the application of the

- This offer will be subject to space availability confirmed by APMTC